

Contents lists available at ScienceDirect

Government Information Quarterly



journal homepage: www.elsevier.com/locate/govinf

Measuring public procurement transparency with an index: Exploring the role of e-GP systems and institutions

Sangeeta Khorana^a, Santiago Caram^b, Nripendra P. Rana^{c,*}

^a Professor and in Endowed chair of International Trade Policy, EFE Aston Business School, Birmingham, United Kingdom

^b Bournemouth University Business School, Bournemouth University, Fern Barrow, Poole BH12 5BB, UK

^c Queen's Business School, Queen's University Belfast, Riddel Hall, 185 Stranmillis Road, Belfast BT9 5EE, United Kingdom

ARTICLE INFO

Keywords:

Transparency

Institutions

Public procurement

Information technology

ABSTRACT

The high expenditure on public procurement by governments makes it imperative to enhance transparency across the procurement cycle with technology-driven initiatives, such as e-procurement systems. This paper develops the Public Procurement Transparency Index and evaluates the impact of institutional reforms and membership of the World Trade Organisation Government Procurement Agreement on transparency. We use the Technology-Organisation-Environment and Balanced e-Participation Index frameworks to analyse transparency in procurement. The Public Procurement Transparency Index uses generalised least squares technique to develop the country-level transparency score. Findings indicate that e-government procurement systems promote transparency, especially in countries with robust institutional frameworks. Further, with fractional probit regression techniques we find institutional quality and infrastructure are key determinants of transparency in public procurement. The results also highlight the importance of information technology and institutional reforms as a means to enhance transparency and accountability in public procurement and offers valuable implications for policymakers.

1. Introduction

Public procurement expenditure accounts for nearly 10-20% of total gross domestic product (GDP) for high-income countries, and this can be as much as 30% in the case of developing countries (Djankov et al., 2017; Knack et al., 2019; World Bank, 2016). Given the high proportion of expenditure there is need to enhance transparency across the entire public procurement cycle, which includes planning, tender, award, contract management and implementation (OECD, 2016). In public procurement, governments are increasingly using e-government procurement (e-GP) systems as a tool to ensure effective use of public funds, reduce fraud and corruption, as well as increase competition among suppliers (Adam et al., 2020; Ghossein et al., 2018; Knack et al. 2019). e-GP systems provide information in open and machine-readable formats that enhances transparency in procurement processes which fosters governments' accountability. At the international level, there are several initiatives to enhance transparency which, among others, include the World Trade Organisation (WTO)-Government Procurement Agreement (GPA) plurilateral agreement. The WTO agreement mandates the use of information and communication technologies (ICT) (e.g., the Internet) and e-GP systems for transparent exchange of information (World Trade Organization (WTO), 2012). Providing access to information and enabling stakeholder participation in the procurement cycle is largely linked to the level of information access allowed and involves political decisions (Ackerman & Sandoval-Ballesteros, 2006; Duguay et al., 2023) which explains why not all countries may have opted for WTO-GPA membership.

Research highlights the role of IT (information technology) infrastructure and the link between digital proficiency, and citizens' ability to interpret and process information in promoting transparency, accountability and citizen participation (see Bertot et al., 2010; Chen et al., 2019; Kabanov, 2022). But digitalisation does not necessarily increase transparency and accountability while reducing corruption (OECD, 2016; Pyman & Heywood, 2021; Schapper, 2008; Seong & Lee, 2004; Ware et al., 2007). In public procurement, e-GP has been proven to ensure accountability and transparency by using integrated e-procurement solutions which increase competition (Ruiz Rivadeneira et al., 2023; World Trade Organization (WTO), 2023). However, with

* Corresponding author. E-mail addresses: s.khorana@aston.ac.uk (S. Khorana), n.p.rana@qub.ac.uk (N.P. Rana).

https://doi.org/10.1016/j.giq.2024.101952

Received 11 September 2023; Received in revised form 5 June 2024; Accepted 9 June 2024 Available online 2 July 2024

0740-624X/© 2024 The Authors. Published by Elsevier Inc. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

increasing integration of IT in governmental processes, the scholarly debate has evolved from questioning what makes policy-making transparent and how to employ transparency as an instrument for effective policy making (Eppel, 2012; Brunswicker et al., 2019; Matheus, Janssen, & Maheshwari, 2020). Attempts have been made to assess transparency in public procurement. Examples include the OECD's (2020a) procurement transparency indices but to the best of our knowledge there is no eprocurement index (i.e. when procurement is conducted using e-GP systems) and how the use of e-GP tools and systems translate into higher transparency. Research shows that an e-procurement transparency index contributes to higher accountability and public trust (Mungiu-Pippidi, 2023) and that having a standardized measure that holds governments and procurement agencies accountable is important. Such tool would also foster fair competition among suppliers, build public trust in government institutions, and serve as a basis for evaluating and improving procurement practices (OECD, 2020). Our paper fills this gap and investigates whether e-GP tools and systems play a role in fostering transparency by examining the link between transparency and institutional reform. This paper firstly, develops procurement-specific measure on country transparency using the Technology-Organisation-Environment (TOE) (Tornatzky & Fleischer, 1990) lens and Balanced e-Participation Index (BEPI) (Pirannejada et al., 2019) framework to offer a comprehensive insight into the dynamics driving transparency in procurement. Secondly, we evaluate the impact of institutional reforms (particularly those related to democracy, government effectiveness, regulatory quality, and contract integrity) on transparency in public procurement. This draws on the broader debate relating to the quality of institutions (Kaufmann, Kraay and Mastruzzi, 2010; Fukuyama, 2013; Rothstein, 2014; Khorana et al., 2014) and examines the impact of WTO-GPA membership, which requires institutional reforms. A related question explored is the role of IT developments (namely e-GP systems) in enabling effective institutions and its relationship with transparency. Data for the Transparency Index is drawn from publicly available sources, namely the World Bank, Organisation for Economic Cooperation and Development (OECD, 2017, 2019a, 2021) and European Research for Anti-Corruption and State-Building (ERCAS). This is complemented by country level data, which is retrieved from individual countries' e-procurement portals. Using a multi-staged sampling process and generalised least squares (GLS) technique, we analyse the transparency score of 133 countries from a total of 218 countries. We draw on institutional variables data from the World Bank Worldwide Governance Indicators (WGI) and employ a fractional probit technique to assess the impact of a country's WTO-GPA membership on transparency in public procurement.

The paper makes two important contributions. Firstly, we develop procurement-specific transparency measure (termed PPT-Index), which analyses whether e-GP tools lead to higher transparency from a policy perspective. Given the high value of public procurement and the general lack of transparency in the process, the unique contribution of the PPT-Index is its ability to foster transparency within the entire e-procurement lifecycle. We develop this index to support governments in assessing the openness, clarity, and the flow of information at each stage of the procurement process. The empirical analysis draws on publicly data which is complemented by country level data to provide policy makers an insight into public procurement related measure for their country. We complement the transparency measurement of the e-procurement cycle with data from bid challenge decisions (complaint mechanism), esignature facility adoption, and access to public information (e-reports, e-statistics, legal gazette, and public official declarations). This also enhances the accountability of the public officials in the procurement processes. Thus, the uniqueness of this paper lies in providing a standardized transparency measure across the entire e-procurement cycle. It also highlights the role e-GP tools play in fostering accountability among governments and public procurement agencies. Second, the paper delves into what role a country's institutional framework plays in ensuring transparency which is essential to increase participation in doing

business with the public sector. For this, we analyse how country-level institutional indicators, which are a part of any country's wider institutions and digital skillset, contribute to transparency in e-public procurement. The research uses data to not only fill a critical gap in existing literature but also highlights the potential of e-GP tools as a means to catalyze competition among suppliers, instill public trust in government institutions, and establish a foundation for transparent procurement practices. Thus, the PPT-Index aims to support governments to meet the policy objectives of transparency and accountability that contribute directly to greater public trust.

The main findings and policy implications are as follows. First, the PPT-Index provides unbiased and evidence-based insights to support informed policymaking in government procurement by emphasising that public procurement practices which foster transparency can progress beyond the abstract notion of 'good' behaviour. Our results confirm that e-GP systems provide access to information on the procurement ecosystem and support transparency in government policymaking. What is evident is that transparency varies across countries with higher transparency in developed countries, especially the WTO-GPA members. Second, we find a direct relationship between transparency and institutional quality (Islam, 2006; Jiménez et al., 2022) and infrastructure (Relly & Sabharwal, 2009) which supports the case for e-GP reforms (Islam, 2003, 2006). Further, the institutional quality of a country matters, i.e., countries with higher government effectiveness, regulatory quality, contract integrity and democratic form of governments, are more likely to have higher transparency in public procurement. The e-GP systems, complemented with digital infrastructure and skills, which provide better access to information support citizens' interaction with the government via web-forms and automated data processing, which in turn promotes transparency and accountability (Correa et al., 2019).

The structure of the paper is as follows: Section two introduces the relevant literature and develops the hypotheses and the theoretical framework. Section three presents the public procurement transparency index and highlights data sources, sampling technique and research methodology. Section four examines the link between country's institutional framework and transparency, discusses the econometric approach, descriptive statistics, and main findings. Section five discusses policy implications. Section six concludes and Section seven highlights direction for future research.

2. Literature, hypothesis development and conceptual framework

2.1. Literature overview and background

The use of e-procurement (also termed e-government procurement or e-GP) has grown in recent years. This form of procurement uses digital technology in procuring and inviting tender for public works, identify potential suppliers of goods and services, to interact with suppliers, to purchase supplies and services in e-marketplaces, as well as for the transfer of payments (Min & Galle, 2003; Standing et al., 2006); and the reduction of delays in the competition of new infrastructure projects by minimising the informational asymmetries, fraud and other prohibited practises in public procurement projects (). Experience from developing countries shows that e-governance can improve transparency which leads to, among others, corruption control and poverty reduction (Bhuiyan, 2011). Along these lines, the decision to adopt egovernment tools also influence the success of new reforms. For instance, the use of software for auctions increases the demand from citizens and firms as well as generates need for higher skills of workers (human capital) to manage new technologies (Aladwani, 2016; Lakka et al., 2013). According to the current research (Janssen et al., 2022; Hujran et al., 2023; Madan & Ashok, 2023), the use of emerging technologies (such as Artificial Intelligence) in innovative procurement improves information flow, support human decisions and allow the acquisition of new hardware and software for different procurement

Table 1

Literature review on e-procurement.

Relevant literature	Focus
Heilbrunn (2004); Seong and Lee (2004); Ware et al. (2007); Hardy and Williams (2008); Schapper (2008); Kim et al. (2009); Pani and Kar (2011); Kim and Lee (2012); Elbahnasawy (2014); Knack et al., 2019; Becker (2018); Schopf (2019); Mélon and Spruk (2020)	Control of corruption, accountability, administrative simplification, integrity transaction costs, policy effectiveness and transparency in procurement process
Relly and Sabharwal (2009); Bertot et al. (2010); Margetts (2011); Gupta and Narain (2012); Cucciniello and Nasi (2014); Jiménez et al. (2022)	Telecommunication, ICT barriers, access to information, free press, social media, digital literacy skills
Larbi, 2007; Ware et al., 2007; Searson and Johnson (2010); Varney (2011); Harrison and Sayogo (2014); Khorana et al., 2015; Lewis-Faupel, Neggers, Olken, and Pande (2016); Chen et al., 2022; Jiménez et al., 2022; Mungiu-Pippidi (2023); Duguay, Rauter, and Samuels (2023)	Legal framework, exchange of complex information, disclosure of public officials, legal repository, norms and regulations

Source: Author's compilation.

stages. In this manner, the use of such technologies not only promotes information transparency, but also enhances accountability, citizen participation and trust in e-government services.

The literature overwhelmingly concurs that the use of e-GP systems increase efficiency through higher competition, transparency, accountability (see for example, Becker, 2018; Henriksen & Mahnke, 2005; Khorana & Kerr, 2021).¹ e-GP tools facilitate an efficient public sector management and lower contract enforcement costs (Gunasekaran & Ngai, 2008; Shim & Eom, 2008). From the bidding entities perspective, e-GP systems facilitates access to a wide pool of vendors and streamlines procurement processes that control spending (Bendoly & Schoenherr, 2005; Moon, 2005), increase entities monitoring capacity through eauctions (Essig & Arnold, 2001; Fish et al., 2015; Rai et al., 2006) and enables the government to work with 'good' (efficient) suppliers whilst decreasing the failure risk of publicly funded projects (Klabi et al., 2018; Vinogradov et al., 2014). In addition, e-GP allows contracting authorities to aggregate demand across departments, reduces inventory costs and overheads (Croom, 2000; Wyld, 2002; Kameshwaran et al., 2007). Thus, the use of e-GP reduces information asymmetry between business with two interlinked effects: first, e-GP systems foster competition and supplier participation in an open market; second, allows procuring agencies to procure quality goods and services at lower prices. The overall efficiency of the e-procurement system, however, depends on the ease of access to information, and processes and procedures (procurement plan, award, etc.) to facilitate wider inclusion of firms in the contracting activity (Pani & Kar, 2011). Table 1 summarises the current state of discussion.

2.2. Hypotheses development

2.2.1. Transparency in public procurement

Transparency has been viewed from several perspectives, such as the release of government information on key economic indicators (Islam, 2006; Roberts, 2006); citizens ability to craft (human capital skills) and demand better and more information (Williams, 2009); openness of government, i.e., the diffusion of information in the public domain (open meetings, decisions). Others focus on citizen monitoring and decision marking (Meijer, 2015; Meijer et al., 2012); public accountability

support (Lourenço, 2015); governance institutions, regulations and credibility (Armstrong, 2005; Ciborra, 2005; Grigorescu, 2003; Otenyo & Lind, 2004; Stasavage, 2003). Yet others highlight the legal environment and documents, laws and/or regulations published (Beblavý et al., 2022) as well as the administrative capacity to implement and monitor policy accumulation to reduce administrative overload (Lourenco, 2023).

From the perspective of public procurement, transparency is based on the belief that democratic governance and efficient markets are enhanced when information is freely available, and administrative agencies have a degree of autonomy, or independence from political interference (Wolfe, 2013). The underlying premise is that transparency can be fostered through national policies, strategy, action plans, dedicated programmes and co-ordination mechanisms that ensure coherence across government departments (horizontally) and different levels of government (national, regional, local) (vertically). In this regard, Lourenço (2023) suggest that a macro perspective of transparency must consider the initial stages of the policy process, agenda-setting and policy formulation. Transparency in public procurement is measured with benchmarks, such as the European Commission's Public Procurement Scoreboard, Open Contracting Partnership, commentaries on the state of regulatory (Public Procurement Act) legal framework in countries (Adam et al., 2020; Correa et al., 2019; World Bank, 2013). However. studies concur that while benchmarking can effectively track and monitor open government initiatives, lack of standardized measures of the progress over time and/or government self-assessment (intentionally or unintentionally) of countries that rank high are attempts to maintain its position rather than improving data provision (Bannister, 2007; Lnenicka et al., 2022; Lnenicka et al., 2024). The WTO-GPA also requires member countries to comply with data reporting requirements, use electronic tools and publish procurement notices to provide equal treatment to domestic and foreign suppliers (Evenett & Hoekman, 2003, 2005; Linarelli, 2003). These create a level playing field for all suppliers so that commitments to create institutions, processes, and decisions are made accessible to the public which allows monitoring, reviewing and commenting on processes and decisions by stakeholders. Enhancing transparency in government procurement features prominently on good governance agenda (Carothers & Gramont, 2011; Križić, 2021). Achieving this agenda necessitates the establishment of 'sound' institutions and implementing measures to facilitate institutional reforms (Khorana et al., 2014; Trepte, 2005; United Nations, 2011).

To foster transparency, countries use e-procurement portals and provide information on procurement lifecycle and frameworks in the public domain, this is linked with the quality of institutions of a country. Institutional reforms can kickstart change, but the 'real' benefits of reform can be reaped by countries only when political bosses recognise the importance of reforms and back them (Szepesi, 2004). There are diverse perspectives on what the quality of institutions means (Fukuyama, 2013; Rothstein, 2014). Kaufmann et al. (2010) offers a broad definition including the modes of access to power, the exercise of power as well as the content of policies. Khorana et al. (2014, p. 122) state "[....] the critical success factors for effective institutional reform are political commitment, the ability to craft an appropriate reform agenda and eventual compliance with rules." The benefits of institutional reform accrue only if the institutions are effective and there is a 'fit' between specific innovations (in this case procurement legislation and e-procurement) and broader institutional environment (Levy & Spiller, 1994). However, the benefits of institutional reform accrue only if the settings are appropriate to kick-start the institutional reform process backed by political leaders' commitment to and willingness to use their weight to support reforms (Khorana et al., 2014, p. 122). Alt and Lowry (1994) argue that democracy increases transparency. In line with the above, we argue that democratic institutions and effective institutions, i. e., functioning e-GP system in a country, support higher transparency which in turn facilitates internal policy coordination accountability in procurement. This leads to the following hypothesis:

¹ Other relevant studies include Cho and Choi (2005); Lenk (2006); Hardy and Williams (2008); Pathak, et al. (2008); Kim et al. (2009); Pathak et al. (2009); Bhuiyan (2011); Varney (2011).

H1. WTO-GPA membership fosters transparency in public procurement.

2.2.2. Link between institutions (i.e., WTO-GPA membership, e-GP) and transparency

Literature highlights the importance of well-functioning institutional frameworks for efficient economic outcomes (North, 1990; Fatas & Mihov, 2013). The implementation of an effective e-GP system enables contracting entities to strengthen competition, transparency, accountability and lowers corruption (see Hardy & Williams, 2008; Kim et al., 2009; Elbahnasawy, 2014; Khorana & Kerr, 2021).

E-procurement systems are, however, linked with the country context which is based on a country's ability to use technology, infrastructure and human capital, and institutions (Heeks, 2005; OECD, 2019; Schapper, 2008). Whitley (1999) argues that e-GP and e-government are components of wider country-level institutional configurations and that e-GP systems cannot be viewed in isolation given these form part of wider institutional arrangements. The implementation of e-GP portals creates an effective institutional framework (Mélon & Spruk, 2020) and e-GP systems foster transparency (Hall & Soskice, 2001). (Fazekas, Lukács, & Tóth, 2015, p. 2) highlight the link between institutional quality and transparency and caution the complexity of the relationship. Studies argue that the political context, technology and wider governance increase citizens readiness to participate in government activities promotes transparency (Jiménez et al., 2022; Pirannejad et al., 2019). Studies confirm that the plurilateral WTO-GPA provides the institutional framework to release procurement related information (Shingal, 2011; Kono & Rickard, 2014; Kutlina-Dimitrova & Lakatos, 2016), and this requires (i) public disclosure of rules; (ii) publication of procurement opportunities; (iii) prior determination and publication of what is to be procured and how submissions are to be considered; (iv) conducting procurement according to prescribed rules and procedures; and (v) how to monitor compliance with rules (World Trade Organization (WTO), 2012). Given that WTO membership requires member countries to have functioning e-procurement portals and creates strong institutional frameworks, the next hypothesis states:

H2. WTO-GPA membership and high institutional quality promote transparency in procurement.

2.2.3. Role of IT in transparency

An institutional framework that prioritises integrity, transparency, and accountability (Fan et al., 2021; Harrison & Sayogo, 2014; Schopf, 2019) requires human capital and technology (Islam, 2003, 2006; Jiménez et al., 2022; Relly & Sabharwal, 2009). Institutional quality must be accompanied by technology and education for effective digital governance (Kim, 2007). To access and utilise the Internet, individuals must have technical skills to use computers, read, comprehend, have ability to search for, use, interpret, and evaluate information (Mossberger et al., 2003, p. 6). The requisite human capital ensures procurement digitisation supports administrative simplification (Jiménez et al., 2022) which contributes to policy effectiveness in the procurement lifecycle, i.e. awards, notifications, bid-challenge recourse, complaint mechanism, e-signature of contracts (Molander, 2014; OECD, 2019a; OECD, 2019b; OECD, 2019c).

The use of e-GP portals relies on Internet speed (Bertot et al., 2010; Meijer, 2013). Studies report that access to telecommunication networks (internet speed, mobile connections) facilitate adoption e-portals, citizens' access to public information (Shapiro, 1999) and improve dissemination of government information while bridging the digital divide in developing countries (International Telecommunication Union (ITU), 2011; Williams et al., 2011). Thus, the digital infrastructure, digital skillset of bidders, entities and citizens are required for an efficient e-GP system (Jiménez et al., 2022). Further, timely access to information coupled with political willingness ensures institutional quality and efficient resource allocation (Williams, 2009; Williams, 2015). This leads to:

H3. : Information technology and WTO-GPA membership impacts transparency in procurement.

2.3. Conceptual framework

The paper draws on two frameworks, namely TOE and BEPI frameworks. TOE considers the interlinkages between technology, organisational factors, and the external environment. Developed by Tornatzky and Fleischer (1990) the TOE framework assesses the adoption of technological innovations, such as IT (Srivastava & Teo, 2010), enterprise resource planning (Haddara & Elragal, 2013; Zhu et al., 2010), and open systems (Chau & Tam, 1997; Wang & Lo, 2016). BEPI (Pirannejad et al., 2019). The BEPI framework evaluates the e-participation initiatives by adding two extensions (namely Human Development Index and Digital Infrastructure) from the society- and government-led initiatives perspectives. This takes into account governmental (top-down) and societal (bottom-up) aspects of e-participation initiatives into account, by focussing on the process of citizen empowerment for engaging in the processes of policy and political decision-making. This dual focus allows for a detailed analysis of procurement processes and highlight areas where citizen engagement and empowerment can be enhanced to improve overall governance effectiveness and inclusivity. Applying BEPI and TOE frameworks to public procurement allows for an in-depth examination of how institutional changes (government-led initiatives) shape procurement policies, and the overall policy formation and uptake of initiatives (society-led) and policy formation in public procurement.

To our knowledge, the TOE and BEPI frameworks have not been used to study the impact of e-GP adoption on transparency. In the context of public procurement, technology refers to the tools and systems used for procurement activities, i.e., e-GP systems, organisational factors include the structure and capabilities of the procurement organisations and officials in a country. The external environment includes factors such as existing market conditions, regulatory requirements, and supplier relationships. By applying the TOE framework, governments can analyse how e-GP impacts overall procurement efficiency, how the organisational structure (which includes e-portals) influence procurement effectiveness, and how external factors (i.e., WTO-GPA membership) shape the overall procurement landscape. Integrating BEPI with TOE provides a structured approach to understand how political, social (suppliers) and technological dimensions together with institutions shape the national public procurement landscape. These enrich the depth and breadth of analysis and provide a robust foundation for analyzing transparency in procurement. An integrated approach enhances the depth and breadth of analysis, and provides a robust foundation to analyse transparency, efficiency, and effectiveness in public procurement.

Fig. 1 presents the conceptual framework which demonstrates the relationship between institutional framework, political willingness, the role of IT infrastructure and literacy in public procurement. we propose that the WTO-GPA encourages countries to use of e-procurement systems which increase efficiency and transparency in public procurement.

We argue that transparency fosters digital citizen empowerment by granting access to government information. Through e-GP portals, which are a part of the broader institutional framework (namely government effectiveness, regulatory environment, contract integrity and democracy), citizens can monitor government actions, contribute to decision-making processes, and hold officials accountable. This is in line with existing literature (Khorana et al., 2014; Fazekas et al., 2015; Pirannejad et al., 2019; Mélon & Spruk, 2020). Trust between governments and citizens is strengthened through transparent communication,



Source: Authors elaboration.



Fig. 2. Institutional Framework and Management Capacity in Procurement. Source: Authors elaboration adapted from MAPS, OECD (2018).

while digital literacy and skills development programs empower citizens to effectively navigate the digital landscape. We posit that political willingness triggers institutional reforms and leads to WTO-GPA membership which in turn fosters transparency (Drabek & Bacchetta, 2004; Allee & Scalera, 2012; Khorana et al., 2014; Choudhury, 2019). The benefits of quality institutions can be harnessed in functioning democracies, where the importance of good institutions is recognised by political leaders. Within this ecosystem, IT plays an important role given the success of e-procurement systems depend on the digital infrastructure and the level of digital literacy of suppliers in the country. It is, thus, a combination of good quality institutions backed by IT infrastructure and digital literacy foster transparency in public procurement.

3. Public procurement transparency index

3.1. Overview

We develop procurement-specific transparency measure, termed PPT-Index, which is a composite index that presents a single score value of transparency. The rationale for developing the index is its usefulness as policy tool to hold governments accountable, to promote competition, and reduce corruption. At present there is a lack of comprehensive eprocurement transparency index, which this paper develops by providing a measurable score for countries. This tool allows procuring agencies and suppliers to monitor fairness and integrity in procurement processes for a particular year. The strength of the Index is its ability to assign individual weights to the different components to provide policymakers at a glance representation of transparency in procurement for effective decision making. The PPT-I enables bidders, stakeholders and citizens to engage with public procurement information from anywhere in the world without going through excessive bureaucratic procedures. According to Lnenicka et al. (2024), access to public data support government transparency actions, creates new opportunities, increases citizen trust in government and reduces information inequalities. Within this context, the PPT-I grants access to government information and aims to "increase the foundation for a well-governed public procurement system", public trust and good governance (OECD, 2018).

The PPT-Index consists of four variables, namely e-GP lifecycle; information on bid challenge decisions; availability of e-signatures for contracts; access to procurement related information (e-reports & estatistics, legal gazette, and public official declarations). The components of the Index draw on the Methodology for Assessing Procurement Systems (MAPS), developed by the World Bank and the Organisation for Economic Cooperation and Development (OECD),² which evaluates how countries' procurement systems work in four main categories, namely Legislative, Regulatory and Policy Framework; Institutional Framework and Management Capacity; Procurement Operations and Market Practices; and Accountability, Integrity and Transparency of the Public Procurement System. Fig. 2 lists variables from OECD MAPS to estimate transparency in e-procurement.

Drawing on Pirannejad et al. (2019), we use BEPI, and assume that eprocurement is a subset of e-participation which allows public (suppliers in this case) to access procurement related information through e-GP systems. The rationale to use BEPI is its ability to emphasise the importance of information flows for higher suppliers e-participation in the government marketplace. BEPI highlights the importance of allowing free access to information for all stakeholders. We hypothesise that free public procurement information is crucial for domestic and foreign suppliers, civil society and the general public. Enabling visibility of the procurement process, from the beginning to the end of the procurement cycle will enhance transparency and accountability. For this reason, we establish the link between the BEPI and e-GP given both share the goal of fostering transparency, efficiency and engagement in government processes through digital technology. In this regard, studies also show that this reduces public funds mismanagement, corruption and increase competition among suppliers to ensure better value for taxpayers' money (Ghossein et al., 2018; Knack et al. 2019).

3.2. Data sources and sampling

Data sources: Fig. 3 presents the modalities to select countries and data sources used for the PPT-Index.

Sampling: Fig. 4 reports the inclusion criteria and the multi-stage sampling methods used to develop the PPT-Index.

list of the countries with links to relevant data sources. Table 2 presents sub indicators and the inclusion criteria of the variables used to develop the PPT-Index.

3.3. Methodology

The PPT-Index is estimated with the following equation:

$$PPT - I = GP_{Cycle} \omega_1 + e - signature \omega_2 + bid - challenge \omega_3 + e - public information \omega_4$$
(1)

where:

 ω are inverse-covariance weights estimated with the generalised least squares (GLS) technique to create a composite index⁴ on transparency in procurement. This was developed by Schwab et al. (2020). This approach does not assume nor impose weighting schemes, thus, weights for each dimension are influenced by the correlation within each subcomponent(s).⁵ Unlike other⁶ weighting techniques, the GLS method is flexible because it assigns lower weights to indicators with missing values and/or high correlated indicators that do not add "new" information to the index. In this way, the summary index reduces mismeasurement errors from the estimation process.

To ensure accuracy, we conduct the correlation of the e-GP cycle with the index components which include e-signature, bid-challenge, e-reports & statistics.⁷ This means if the correlation between sub-components is high, the GLS assigns a lower weight to variables which is likely to affect the distribution of other sub-indicators of the index. A tetrachoric factor analysis is employed and a new component, namely "e-public information" which contains e-reports, e-statistics, legal gazette, and disclosure of public officials is created to reduce noise between the variables. Kolenikov and Angeles (2004, 2009) show that the conversion of categorical (ordinal) into indicator (binary) variables can modify the orientation of the axes and introduce spurious correlations. We perform an oblique rotation, wherein factors are moderately intra-

Following a staged sampling approach, we collect information on 218 countries in stage 1. Stage 2 excludes 15 countries with missing information, and the sample is reduced to 203 countries. Stage 3 checks data from the (OECD, 2017 2019, OECD, 2021). Stage 4 uses secondary data from the European Research for Anti-Corruption and State-Building (ERCAS) published in 2022, this has 129 countries. Finally, stage 5 manually gathers missing data from individual country portals. The final dataset includes 133 countries. Table A.1 presents the final list of countries included for analysis in the sample.³ Annex A presents the full

² MAPS is an OECD tool that aims to "increase the foundation for a wellgoverned public procurement system", Details of the methodology can be accessed at https://www.oecd.org/gov/public-procurement/methodolo gy-assessing-procurement/

³ Overall, 85 countries were removed from the original sample due to lack of data for all variables.

⁴ Several weighting methods are used to construct a composite index such as the equal weights method, this is used for ethics, open budget, public integrity, corruption and transparency (Bandura, 2008; Mungiu-Pippidi, 2023; Mungiu-Pippidi & Dadašov, 2016; OECD, 2008). Partially non-compensatory technique of geometric aggregation i.e. square root of the product between each dimension, has been used by OECD (2008). Lidén (2015) and Kabanov (2022) employ the refined e-participation index to consider democracies and citizen engagement in public policymaking. The best-worst approach uses a structured pairwise comparison method to account for varying importance of items in an aggregate index (Rezaei, 2015; 2016; Pirannejad et al., 2019). Finally, the unobserved components model (UCM) aggregates multiple sources of data into a single composite index, e.g. WGI indicators, assumes that the error within each source is uncorrelated across other sources and the variance across respondents remains the same (Kaufmann et al., 2006; Kaufmann et al., 2015).

⁵ Some empirical issues are worth stressing. First, if components are highly correlated weights can be zero or negative. Second, information on individual variables is likely to be cross reported, in which case the GLS would increase random errors of variables which are correlated with other components (Schwab et al., 2020).

⁶ Equal weights and non-compensatory schemes have strong assumptions and neglect the individual value of each component and/or possible interactions mechanisms which is a strong and normative assumption and possible interactions between components. UCM extracts common factors from different sources but assumes that weights are proportional to the reliability of each source. This means that each source is uncorrelated across countries., Hence, we use the summary index approach to calculate weights with the GLS procedure (Anderson, 2008; Schwab et al., 2020) which accommodates to the data generating process without imposing equal weights, minimizes standard errors of uncorrelated variables and also handles indicators with missing data.

⁷ For procurement reports and statistics, a significant and positive correlation (0.729) is documented whereas the other variables, namely e-signature and bid-challenge exhibit a positive but moderate correlation.



Fig. 3. Data sources. Source: Authors compilation.

correlated compared to orthogonal rotations, to preserve the norms of the loading matrix rows (Clarkson & Jennrich, 1988). The components with an eigenvalue higher than 1 are retained.⁸ Thus, a single factor contains information from the three sub-indicators on e-public information dimension, which excludes potential spurious correlation. This new variable is normalized to fit the range from 0 to 1.⁹ Finally, to facilitate transparency comparison between countries the PPT-Index is rescaled¹⁰ with values between 0 and 1, where 0 is "completely opaque" and 1 "fully transparent". a score of 1.¹¹ Fig. 5 presents the weights of the PPT-Index and Table A.2 (Panel B) reports the Cronbach's alpha of the dimensions of transparency in public procurement. Results indicate that the PPT-I is internally consistent with an average Cronbach alpha of 75% which is acceptable.

Fig. 5 shows e-GP cycle contributes 39% on average to transparency, followed by bid-challenge (29%) and e-signatures (22%). *E*-public information has a smaller share, only 10%.

The results are fitted into Eq. (1) as below:

$$\begin{split} \text{PPT}-\text{I} &= \text{GP Cycle } \omega_{(0.389)} + \text{e} - \text{Signature } \omega_{(0.290)} \\ &+ \text{bid} - \text{challenge } \omega_{(0.219)} + \text{e} - \text{public information } \omega_{(0.103)} \end{split}$$ $\end{split} \tag{2a}$

The estimated weights show that the e-GP cycle and bid-challenge recourse constitute 68% of the PPT-Index which highlights the importance of procurement information dissemination before the contract is awarded. This is to reduce the risk of corruption and increase transparency (Bauhr et al., 2020; Knack et al., 2019) and public scrutiny of procurement officials from investigative journalists (Chen & Neshkova, 2020), government watchdogs and regulators through more competitive and monitoring procedures (Duguay et al., 2023).

3.4. Stylized facts

Table 3 presents a snapshot of transparency from the PPT-Index in public procurement by region.

The global score base of the PPT-Index is 0.705.¹² We observe that North America and European countries rank first with an average transparency of 0.955 and 0.906 while Sub-Saharan Africa is at the bottom of the list with 0.528. In the case of South America, transparency scores are also lower (0.577) compared to Asia (0.752).For instance, within countries transparency scores range from 0.748 up to 1 for North America (USA, Canada, Mexico), Europe, Latin America (Chile, Peru) and East Asia and Pacific (Korea, Australia, Japan, Singapore, New Zealand). Exceptions include China, Malaysia, Mongolia, Philippines, and Timor-Leste, where the scores are between 0.600, Cambodia and Myanmar with a score close to zero (0.037). Similar scores are observed for Sub-Saharan African countries (0.563), Middle East and North Africa

⁸ Table A2 (panel a) reports the factor loadings. The retained factor namely "e-public information" is standardized and rescaled from 0 to 1.

⁹ A similar procedure was also applied by Pirannejad et al. (2019) and Kabanov (2022).

¹⁰ It is worth mentioning that the PPT-Index is continuous variable whilst all sub indicators with the exception of the e-public information index are binary variables (0 or 1). This differentiation is necessary because all normalisation methods which are invariant to the interval scale, are also invariant on the ratio scale (OECD, 2008, p. 85).

 $^{^{11}}$ This means that countries that countries with the lowest score across four dimensions of procurement (e.g. Namibia -2.8) are assigned a score of 0. Conversely, countries with robust e-procurement systems (e.g. United Kingdom 1.17).

¹² The average global transparency sub-score estimated by Mungiu-Pippidi, 2023 is 0.697 but this is based on procurement awards and notifications and does not include the full procurement cycle.



Fig. 4. Inclusion criteria for PPT-Index. Source: Authors compilation.

(0.578), Latin America (0.609) and Central America with the exception of Belize and Barbados which show lack of transparency in procurement. Exceptions include Peru and Chile (0.793, 0.748) in Latin America. In sub-Saharan Africa these include Rwanda (0.748), Cameroon, Ethiopia, Zambia (0.711); Malta, Jordan, Qatar (Middle East and North Africa) where procurement transparency ranges from 0.600 to 0.955.

Fig. 6 shows the average transparency score by WTO-GPA membership, WTO-GPA Observers and WTO-GPA non-members.

Transparency levels exhibit heterogeneity across countries and regions. The results show that the WTO-GPA parties report a higher transparency in procurement (0.969). For Observer countries the transparency score is 0.676. For non-GPA members, the score is 0.528, which is 83.5% lower than WTO-GPA members. The explanation for varying levels of transparency is due to difference in the level of information provided by governments and also because some countries are WTO-GPA members and other are not. This follows from the binding commitment of WTO-GPA countries to disclose public procurement information throughout the entire lifecycle. However, the quantity of information disclosed may be influenced by factors such as e-portals maintenance, institutional quality, and digital infrastructure networks, leading to varying levels of transparency across countries. Transparency in procurement is further complicated by the political nature of government decisions and institutional factors. The WTO-GPA mandates for signatories to disseminate procurement data translates into benefits for

businesses seeking to participate in international procurement by providing greater access to information, thereby increasing competition and fairness. However, it poses a challenge for governments with limited digital infrastructure or weaker institutions, as they may struggle to meet these transparency requirements. Consequently, we suggest that while WTO-GPA membership provides a necessary condition for enhancing transparency, it may not be sufficient without the abovementioned conditions.

Fig. 7 displays the correlation for transparency by WTO-GPA and non-WTO-GPA members.

We observe that legal mechanisms, public information dissemination, e-signature technologies and e-GP cycle differ by WTO-GPA membership. For WTO-GPA signatory countries the correlation is positive and highly significant; for WTO-GPA non-members the effect is negative and statistically significant at the 1% level. The positive correlation for GPA countries indicates that compliance with WTO GPA rules directly correlates with higher transparency levels. Further, legally binding measures on sharing procurement information through e-portals are a driver for higher transparency in WTO-GPA member countries. The explanation for higher score for GPA members is that member countries are obliged by the WTO-GPA to adopt transparency provisions and e-GP systems, which we test in Section 4.

Table 2

Inclusion Criteria.			
Dimension	Subcomponent	Inclusion Criteria	Inclusion Code
e-GP cycle	e-publishing	e-procurement platform publishes procurement plans of procurement entities	A score of 1 is assigned if this feature is available in a country
	Online documents	e-platforms allow bidders to download tender documents.	A score of 1 is assigned if this feature is available in a country
	Awards	e-platforms publishes contract award decisions	A score of 1 is assigned if this feature is available in a country
Bid challenge/Complaint mechanism	Bid challenge	Unsuccessful bidders can challenge contract award decisions through the country's portal.	A score of 1 is assigned if this feature is available in a country
e-signature	e-signature	The country provides advanced e-signature and other authentication technologies in procurement contracts	A score of 1 is assigned if this feature is available in a country
Public information on e- procurement	Online reports	Government regularly publishes and provides access to online reports on procurement contracts	A score of 1 is assigned if this feature is available in a country
	Online statistics	Government regularly publishes and allows access to online statistics on procurement contracts	A score of 1 is assigned if this feature is available in a country
	Legal gazette	The country provides information on procurement regulations (and changes) through the official legal gazette	A score of 1 is assigned if this feature is available in a country
	Conflict of interest	The country provides information on disclosure of public officials	A score of 1 is assigned if this feature is available in a country





Fig. 5. Public Procurement Transparency Index (PPT-Index) - weight of the components.

(Source: Authors elaboration based on GPPD, OECD and country e-portals.)

Table 3

Transparency in procurement by region.

Region	PPT-I (average score)	Number of countries
North America	0.955	2
Europe and Central Asia	0.906	46
South Asia	0.752	5
East Asia and Pacific	0.669	15
Middle East and North Africa	0.597	12
Latin America and the Caribbean	0.577	26
Sub-Saharan Africa	0.528	27
World	0.705	133

Source: Authors compilation.

4. Examining the link between institutional framework and information technology

Governments play a critical role in promoting transparency by allowing access to information on the rules, regulations and the overall regulatory framework. When bidders have access to information that is complete and transparent information, they can exploit the procurement possibilities. A strong institutional framework that promotes transparent systems with rules and regulations acts as an incentive to bidders and increases competition in the market, which delivers value for money.



Fig. 6. Average transparency score for WTO-GPA, WTO-Observers and WTO non members.

Source: Authors elaboration.



Fig. 7. Correlation of PPT-Index components for WTO-GPA and non-GPA members.

Note: ***, **, * indicate statistical significance at 1, 5 and 10% level, respectively.

Source: Authors compilation.

The underlying reason for providing information is that informing the bidders about the procurement institutional framework ensures efficient resource allocation (Williams, 2009; Williams, 2015; Fan et al., 2021). But quality institutions must be complemented with IT access i.e., digital infrastructure and literacy, to facilitate efficient procurement systems that drive efficiency gains for contracting entities, bidding firms, tax-payers (Jiménez et al., 2022; OECD, 2019b; OECD, 2019c).

Table 4

Description of variables, definition and sources.

Variable	Definition	Data source
PPT-Index	Public procurement transparency index	Own compilation
e-GP	Equal 1 if the country introduced e-GP tools, 0 if not	Government Public Procurement Database (GPPD) and OECD
GPA	Equal 1 if the country signed the GPA agreement, 0 if not	World Trade Organisation (WTO)
Observer	Equal 1 if the country is a WTO observer, 0 if not	World Trade Organisation (WTO)
Non- member	Equal 1 if the country is not a WTO member, 0 if not	World Trade Organisation (WTO)
OECD	Equal 1 if the country is an OECD member, 0 if not	Organisation for Economic Co-operation and Development (OECD)
dem_pol_will	Quality of democracy, political interests, accountability human heights and freedom of association (EIU) in 2021	World Governance Indicators (WGI)
inst_effect	Institutional and government effectiveness (EIU) in 2021	World Governance Indicators (WGI)
reg_envir	Regulatory environment, compliance and bureaucratic inefficiency (WMO) in 2021	World Governance Indicators (WGI)
contract_int	Contract integrity, securing major contracts from risk of bribery (WMO) in 2021	World Governance Indicators (WGI)
i_band_speed	International bandwidth per Internet user (bit/s) in 2021	International Telecommunication Union (ITU)
int_band_s	Fixed broadband subscriptions (per 100 people) in 2021	World Development Indicators (WDI)
mob_band_s	Mobile cellular subscriptions (per 100 people) in 2021	World Development Indicators (WDI)
access_elec	Access to electricity (% of population) in 2021	World Development Indicators (WDI)
digital_lit	Ability to use digital technology of all parts of society in 2022	Digital Development Compass (UNPD)
digital_infra	Digital infrastructure index	Own compilation

Source: Authors compilation.

4.1. Data

Public procurement related data is taken from the World Bank (GPPD) and OECD (2017, 2019a, 2021). Transparency related institutional variables are from World Bank's Worldwide Governance Indicators (WGI) (see Kaufmann et al., 2010; Kaufmann and Kraay, 2023). We consider four WGI sub-variables, namely 'government effectiveness', 'regulatory environment', 'contract integrity' and democracy (proxy for political willingness)'.¹³ We rationalise that procurement as government activity is linked with public policy effectiveness (Khorana et al., 2014) and contract security which are both related to democracy.¹⁴ Data on internet speed users is from the International Telecomm Union (ITU); while broadband internet, mobile subscriptions and access to electricity is from the WDI (World Bank, 2023). Data on digital literacy (ability to use digital technology of all parts of society) is from the United Nations Development Program (UNDP). Table 4 lists variables, definitions and data sources. Table A.3 outlines the datasets, variables used and data links.

We construct procurement-specific digital infrastructure index using the GLS approach 15 with:

i) international internet speed users, bits per second

ii) broadband internet subscriptions, per 100 people

- iii) mobile subscriptions, per 100 people
- iv) access to electricity (percentage of population).

The weights, assigned with the GLS procedure, show that broadband accounts on average 41%, internet speed 33%, mobile subscriptions 16% and access to electricity 11%.

 $\mathrm{D-I} = \mathrm{broadband} - \mathrm{subscriptions} \ \omega_{(0.406)} + \mathrm{internet} \ \mathrm{speed} \ \omega_{(0.330)}$

+ mobile – subscriptions $\omega_{(0.156)}$ + access electricity $\omega_{(0.108)}$

(2b)

¹⁵ The digital infrastructure index does not employ factor analysis approach due to low interterm correlations.

4.2. Econometric approach

The dependent variable, PPT-Index, is continuous with a closed bounded domain [0,1]. Given the predicted values of linear models may lie outside the unit-interval, we employ a fractional probit technique, which is in line with Papke and Wooldridge (1996) and Wooldridge (2010). We estimate H1, H2, and H3 with following equations:

H1. WTO-GPA membership, supported by a strong political willingness to reform and complemented by e-GP systems, fosters transparency in public procurement.

$$y_{t} = \beta_{0} + \beta_{1}eGP_{i} + \beta_{2}WTOGPA_{i} + \beta_{3}WTOGPA_{OBi} + WTOGPA_{i} \times \beta_{4}dem_{t} + WTOGPA_{OBi} \times \beta_{5}dem_{t} + \epsilon_{i}$$
(3)

H2. WTO-GPA membership together with e-procurement portals and strong institutional frameworks promotes transparency.

$$y_{t} = \beta_{0} + \beta_{1}eGP_{i} + WTOGPA_{i} \times \beta_{2}instqlty_{t} + WTOGPA_{OBi} \times \beta_{3}instqlty_{t} + \epsilon_{i}$$
(4)

H3. Transparency depends on both WTO-GPA membership and the level of information technology (digital infrastructure and digital literacy) available in a country.

$$y_{t} = \beta_{0} + \beta_{1}eGP_{i} + WTOGPA_{i} \times \beta_{2}DI_{t} + WTOGPA_{OBi} \times \beta_{3}DI_{t} + WTOGPA_{i} \times \beta_{4}dl_{t} + WTOGPA_{OBi} \times \beta_{5}digt_{t} + eGP_{i} \times \beta_{6}DI_{t}$$
(5)
+ $eGP_{i} \times \beta_{7}dl_{t} + c_{i}$

where

 y_t is the transparency index in public procurement (PPT-Index); eGP_i is the dummy for active e-GP portals;

*WTOGPA*_i is the dummy for WTO-GPA membership;

WTOGPA_{OBi} is the dummy for WTO-GPA Observer status;

WTOGPA_{NONi} is the omitted category (non-membership to WTO-

GPA);

*dem*_t is a measure of democracy (political willingness);

 $instqlty_t$ is a vector of institutional indicators, namely government effectiveness, regulatory environment and contract integrity;

¹³ We use the (standardized) estimated parameters from the UCM unlike the aggregate institutional indicators.

¹⁴ As Williams (2011, p.813) states, "the country scores can be used to compare relative governance across countries in time t, but the scores for each country for time t, t + 1, t + 2 etc. are not directly comparable". Hence, given the cross-sectional nature of our study, we use the estimated parameters instead of the aggregated WGI scores.

 DI_t is the digital infrastructure index;

 dl_t is the level of digital skills in a country;

Subscript i is reported for dummy variables, t is time which is the latest year of data available and e_i is the error term.

The fractional probit approach provides a consistent estimation, unlike the ordinary least squares (OLS) where the predicted¹⁶ values of linear models "[...] can never be guaranteed to lie in the unit interval" Papke & Wooldridge (1996, p. 620)." We also consider the following methodological issues. First, the empirical analysis uses cross-sectional data. This allows running pooled models. Second, the potential endogeneity of a covariate is addressed with an interaction between the endogenous variable(s) and exogenous binary indicator(s); the standard statistical inference for the interaction terms applies (see Nizalova & Murtazashvili, 2016; Beverelli et al., 2018; Bun & Harrison, 2019). The WTO-GPA and WTO-GPA Observer countries are interacted with institutional quality variables, digital infrastructure, and digital literacy. The interactions show an indirect effect of the WTO-GPA membership and Observer status on transparency through political willingness, institutional quality, information technology. Third, to avoid multicollinearity issues we assess the indirect effects of institutional and digital variables with one indicator at a time.¹⁷

4.3. Descriptive statistics and discussion of results

Table 5 presents the PPT-Index and digital index descriptive statistics by WTO-GPA membership.

WTO-GPA parties have higher PTT-Index and digital infrastructure (0.567) compared to WTO-Observers (0.367). Digital literacy is high for GPA countries (0.904). For instance, from GPA parties, Korea, Switzerland, France, Estonia and the UK have highest access to digital infrastructure (internet speed, mobile connections), 0.647 on average. A similar pattern is observed for digital literacy, which is 0.912 on average, for these countries.

WTO-GPA Observers and GPA non-members have low digital infrastructure and digital literacy, this is likely to affect the e-procurement portal usage. Examples of WTO-GPA Observers with e-procurement portals are Kyrgyz, Georgia and Thailand with digital infrastructure and literacy levels, at 0.406 and 0.833 average, respectively.

For WTO-GPA non-members (e.g. Angola, Lesotho, Zimbabwe), IT infrastructure (internet speed, mobile connections) and digital literacy is lower (0.149 and 0.570) on average. For countries, such as Sierra Leone and Chad, which have no e-procurement portal also have limited access to digital infrastructure (0.070) and digital literacy (0.215). This is not surprising given most WTO-GPA non-member parties are developing countries with limited access IT infrastructure (0.260) and digital literacy capacities (0.600). Table 6 presents the top 20 and bottom 20 countries in the order of ranking.

Results confirm that transparency is lower for countries that do not mandate to comply with GPA commitments. The ranking for all countries in our sample is reported in Table A.4. We observe that the institutional frameworks affect transparency in public procurement. For instance, the WTO-GPA Parties exhibit high institutional quality as evidenced by government effectiveness (0.718), regulatory environment (0.734), contract integrity (0.733) and democracy (0.754). But this is not the case for WTO-GPA non-member parties where the quality of institutions, namely government effectiveness (0.221), regulatory environment (0.460), contract integrity (0.353) and democracy (0.387), Table 5

Descrip	tive s	tatis	tics.
---------	--------	-------	-------

Panel A: WTO-GPA party	Mean	SD	Min	Max
PPT-Index	0.969	0.037	0.852	1
Democracy (political willingness)	0.754	0.128	0.461	0.956
Government effectiveness	0.718	0.217	0.250	1
Regulatory environment	0.734	0.108	0.500	1
Contract integrity	0.733	0.189	0.333	1
Digital infrastructure	0.567	0.097	0.398	1
Digital literacy	0.904	0.051	0.790	1
Number of countries	43			
Panel B: WTO-GPA Observer status				
PPT-Index	0.676	0.077	0.560	0.793
Democracy (political willingness)	0.454	0.161	0.128	0.740
Government effectiveness	0.415	0.175	0	0.750
Regulatory environment	0.546	0.152	0.250	0.917
Contract integrity	0.409	0.148	0.167	0.833
Digital infrastructure	0.367	0.089	0.151	0.540
Digital literacy	0.765	0.116	0.390	0.920
Number of countries	31			
Panel C: WTO-GPA non-member				
PPT-Index	0.528	0.238	0	1
Democracy (political willingness)	0.387	0.178	0.051	0.849
Government effectiveness	0.221	0.192	0	0.625
Regulatory environment	0.460	0.159	0	0.750
Contract integrity	0.353	0.186	0	0.833
Digital infrastructure	0.260	0.129	0	0.560
Digital literacy	0.600	0.170	0.180	0.900
Number of countries	59			

Source: Authors compilation.

is likely to lead to a lack of transparency. This is because a poorly implemented institutional framework influences a society's incentive system, which in turn affects social interactions and individual's behaviours (North, 1990). Taken together, the results confirm that transparency in procurement is likely to be determined by a country's WTO-GPA commitments, political willingness, institutional frameworks, digital infrastructure and literacy.

Tables A.5 and A.6 present pairwise correlation and multicollinearity analysis. The mean of Value Inflation Factor (VIF) is 4.773 which suggests no multicollinearity below the rule of thumb of 10.

4.4. Discussion

H1. WTO-GPA membership fosters transparency in public procurement.

Table 7 (column I) reports WTO-GPA membership fosters transparency in public procurement. The GPA parties present evidence of higher transparency (1.623) with positive and statistically significant coefficient at the 1% level. The positive and significant relationship between e-GP adoption and transparency (2.391) at the 1% level suggests WTO-GPA parties have institutional frameworks for information on tender notifications, contract awards, and e-quotations. WTO-GPA Observers, with no mandate to comply with GPA requirements, have lower transparency (0.212) but it is highly significant at the 5% level.

Column II reports results when democracy (political willingness) is introduced. For GPA Parties democracy is positive (2.058) and statistically significant at the 1% level, suggesting political willingness to reform leads countries to join WTO-GPA membership which fosters transparency. For WTO-GPA Observers, the impact on transparency is 0.325 and significant at the 5% level suggesting democratic government are likely to publish data which fosters transparency and citizens trust in e-government services. In line with extant literature on e-government initiatives, political commitment, e-participation and new mechanisms to increase transparency (Bisogno et al., 2022; Cinar et al., 2019; Cinar et al., 2021; Moser-Plautz, 2023; Savoldelli et al., 2014). In this case

¹⁶ OLS can be augmented with a non-linear function of the independent regressor(s), but these functions must include the predicted values from the linear regression model into the equation. Papke and Wooldridge (1996) provide a detailed explanation of the shortcomings of this procedure.

¹⁷ For instance, the WGI are likely to be highly correlated because of measurement errors from survey sources and the assessment given to each country (Kaufmann et al., 2007).

S. Khorana et al.

Table 6

Тор	and	bottom	20	countries	with	transparency	in	public
proc	ureme	ent.						

WTO-GPA member	PPT-I (overall score)
Armenia	1
Bulgaria	1
Croatia	1
Estonia	1
Finland	1
France	1
Greece	1
Israel	1
Italy	1
Korea, Rep.	1
Latvia	1
Lithuania	1
Moldova	1
Portugal	1
Romania	1
Slovak Rep.	1
Spain Switzenland	1
Juraino	1
United Kingdom	1
WTO-Observer	PPT-I (overall score)
Georgia	0 793
Thailand	0.793
Albania	0.748
Chile	0.748
India	0.748
Indonesia	0.748
Jordan	0.748
Kyrgyz Republic	0.748
North Macedonia	0.748
Russia	0.748
Sri Lanka	0.748
Turkey	0.748
Vietnam	0.748
Cameroon	0.711
Kazakhstan	0.711
Pakistan	0.708
Argentina	0.645
Paraguay	0.645
Dominican Rep.	0.608
Colombia	0.608
WTO non-member	PPT-I (overall score)
South Africa	0.563
Tanzania	0.563
Uganda	0.563
remen	0.563
DUIIVId	0.560
II dii Venezuelo	0.560
v cu czuela Grenada	0.500
Lesotho	0.523
St Lucia	0.523
Zimbabwe	0.523
Algeria	0.037
Cambodia	0.037
Myanmar	0.037
Barbados	0
Belize	0
Chad	ů 0
Lebanon	0
Namibia	0
Sierra Leone	0

Source: Authors compilation.

countries that signed for the GPA are more likely to report high transparency levels in procurement.

H2. WTO-GPA membership and high institutional quality promotes transparency.

Table 7 (columns III-V) shows transparency is impacted indirectly by institutional quality through WTO-GPA membership. The impact of GPA

membership on transparency is evidenced by the coefficients of government effectiveness (2.089), regulatory environment (2.139) and contract integrity (2.058), and all the interaction terms are statistically significant at the 1% level. The GPA parties have active e-procurement portals when complemented with robust institutions foster on average transparency (2.429). For Observer countries, the effect of institutional quality on transparency is modest but statistically significant at the 5% level. Point estimates confirm institutional quality, namely government effectiveness (0.334), regulatory environment (0.358) and contract integrity (0.379), have a moderate impact on transparency.

The is because firstly, the GPA requires providing equal and fair access to procurement information and mandates use of e-GP (World Trade Organization (WTO), 2012). Secondly, institutional quality (government effectiveness, regulatory environment, contract integrity) is influenced by political willingness to reform (Citro et al., 2021; Khorana et al., 2014) and government openness trust in the public sector and its institutions (Schmidthuber et al., 2021). Along these lines, we argue that a combination of effective institutions from WTO-GPA membership enhances information to the bidders on the e-GP lifecycle, bid-challenge, e-signature, e-public information. This improves accountability, increases trust in e-government applications accountability and lowers the transaction cost (Barcevičius et al., 2019; Bauhr & Grimes, 2014; Bosio et al., 2022; Kim & Lee, 2012; Knack et al., 2019).

H3. Information technology and WTO-GPA membership impacts transparency in procurement.

Table 8 (Column I) shows that IT infrastructure is an enabling factor, given WTO-GPA membership mandates countries to use e-GP tools. The digital coefficient for the GPA parties is positive (2.748) and statistically significant at the 1% level whereas for WTO-Observers the effect is 0.488 and significant at the 5% level. Results highlight the importance of adequate digital infrastructure and the dissemination of public information released by governments for transparency (Bisogno et al., 2022; Lathrop & Ruma, 2010; Noveck, 2009; Relly & Sabharwal, 2009; Twizeyimana & Andersson, 2019).

Column II shows the positive effect of the WTO-GPA membership and digital skills on transparency is 1.779 and statistically significant at 1% level, unlike WTO-GPA Observers which is 0.268 and significant at 5% level. The coefficient for the e-GP system is 2.404 on average and highly significant, suggesting that improvements in information technology, such better access to high-speed broadband internet and mobile networks in conjunction with digital literacy capabilities, support higher transparency procurement. Results confirm the importance of skills (ecompetences) to use e-government systems and understand the information, e-participation in the decision-making process and government accountability to increase transparency (Chen & Neshkova, 2020; Evans & Yen, 2005; Kearns, 2004; Zheng, 2016).

Columns (III-IV) report the indirect effect of e-GP systems through digital infrastructure and digital literacy on both WTO-GPA and WTO-GPA Observer countries. For WTO-GPA parties, digital infrastructure and literacy skill remain positive (2.125 and 1.393) and highly significant. In comparison, for WTO-GPA Observers, the coefficient is positive (0.291) but statistically insignificant. The same pattern is observed for digital literacy where the coefficient is 0.047 and insignificant. Interestingly, the interaction between e-procurement portals, digital infrastructure and digital literacy is positive and highly significant, suggesting that countries that provide better connectivity services (broadband internet, mobile) with digital literacy increase transparency by 1.581 and 1.435, respectively.

Taken together, the results reinforce the findings that high quality of digital infrastructure and digital literacy of e-procurement portals leads to higher transparency. In line with existing literature (Barcevičius et al., 2019; Bosio et al., 2022; Jiménez et al., 2022; Khorana et al., 2014) when procurement under the GPA rules is complemented with digital infrastructure and literacy skills, citizens' trust in e-portals and overall transparency increases.

Table 7

Transparency in Procurement, WTO-GPA membership and Institutional quality.

Variables \setminus Model specification	(I)	(II)	(III)	(IV)	(V)	
	Dependent variable: PPT-Index					
e-GP	2.391*** [0.183]	2.429*** [0.183]	2.450*** [0.183]	2.405*** [0.183]	2.433*** [0.183]	
WTO-GPA	1.623***	[0.100]	[0.103]	[0.100]	[0.100]	
	[0.096]					
WIO-GPA Observer	0.212** [0.066]					
WTO-GPA \times Democracy (political willingness)	[01000]	2.058***				
		[0.147]				
WTO-GPA Observer × Democracy (political willingness)		0.325**				
WTO-GPA \times Government effectiveness		[01107]	2.089***			
			[0.176]			
WTO-GPA Observer \times Government effectiveness			0.334**			
WTO-GPA \times Regulatory environment			[0.111]	2.139***		
				[0.149]		
WTO-GPA Observer \times Regulatory environment				0.358**		
WTO-GPA \times Contract integrity				[0.110]	2.058***	
					[0.176]	
WTO-GPA Observer \times Contract integrity					0.379**	
Constant	-2.145***	-2.145***	-2.145***	-2.145***	-2.145***	
	[0.175]	[0.175]	[0.175]	[0.175]	[0.175]	
Pseudo R-squared	0.243	0.239	0.232	0.237	0.230	
Log-Likelihood	-61.070	-60.197	-60.780	-61.521	-62.126	
Number of countries	133	131	131	133	133	

Note: e-GP, WTO-GPA and WTO-GPA Observers are binary variables (0 or 1). Robust standard errors in brackets, ***;**,* indicate statistical significance at 1, 5 and 10% level, respectively.

Source: Authors compilation.

Table 8

Transparency in Procurement, WTO-GPA membership and Information technology.

Variables \ Model specification	(I)	(II)	(III)	(IV)
	Dependent v	ariable: PPT-In	dex	
e-GP	2.412***	2.395***		
	[0.183]	[0.183]		
WTO-GPA \times Digital	2.748***		2.125***	
infrastructure				
	[0.214]		[0.394]	
WTO-GPA Observer \times	0.488**		0.291	
Digital infrastructure				
	[0.174]		[0.296]	
WTO-GPA \times Digital		1.779***		1.393***
literacy				
		[0.109]		[0.187]
WTO-GPA Observer \times		0.268**		0.047
Digital literacy				
		[0.085]		[0.150]
e -GP $^+ imes$ Digital			1.581**	
infrastructure				
			[0.597]	
e-GP \times Digital literacy				1.435***
				[0.367]
Constant	-2.145^{***}	-2.145^{***}	-0.267	-0.697***
	[0.175]	[0.175]	[0.166]	[0.212]
Pseudo R-squared	0.236	0.241	0.183	0.205
Log-Likelihood	-61.643	-61.178	-65.860	-64.087
Number of countries	133	133	133	133

Note: e-GP, WTO-GPA and WTO-GPA Observers are binary variables (0 or 1). Robust standard errors in brackets, ***;**,* indicate statistical significance at 1, 5 and 10% level, respectively.

Source: Authors compilation.

Fig. 8 presents the relationship between institutional framework, political willingness and IT (digital infrastructure and digital literacy). Results confirm political willingness, proxied by democracy, triggers institutional reforms and membership to the WTO-GPA which in turn leads to effective e-procurement portals use. Thus, sound institutions (e. g. contract integrity) and better access to digital infrastructure resources (connectivity and speed of broadband internet, mobile services) and the level of digital literacy of citizens backed by political willingness and IT initiates a virtuous cycle of procurement transparency.

4.5. Robustness checks and sensitivity analysis

To test the robustness, we firstly assess whether results hold for OECD countries. Table 9 (Column I) shows that transparency increases by 1.140 in OECD countries compared to non-OECD countries. The e-GP coefficient is positive (2.555), and statistically significant at the 1% level. In addition, democracy (Column II) increases PPT-Index by 1.516 and institutional variables, namely, government effectiveness (1.559), regulatory environment (1.680) and contract integrity (1.530) are highly significant (columns III–V). These confirm that political leadership and institutional frameworks are important drivers of e-GP system and transparent procurement.

Table 10 (columns I-IV) show that OECD countries have higher IT infrastructure (2.221) which when complemented with digital literacy of suppliers (1.274) fosters transparency in public procurement (columns I-IV).

The use of e-GP by OECD countries increases transparency (2.365) and when countries provide better access to IT infrastructure (internet speed, broadband connectivity mobile access) with high digital literacy transparency increase is 1.829. These findings, are expected as developed countries invest more in infrastructure and hence, have more resources for e-government initiatives which in turn increase e-participation and transparency. However, e-procurement systems



Note: Authors elaboration, ***;**,* indicate statistical significance at 1, 5 and 10 percent level, respectively

Fig. 8. WTO-GPA membership, Institutional quality, Information technology and Transparency in Public Procurement.

Table 9

Transparency in Procurement, OECD membership and Institutional quality.

Variables \ Model specification	(I)	(II)	(III)	(IV)	(V)		
	Dependent variable: PPT-Index						
e-GP	2.555***	2.563***	2.563***	2.550***	2.560***		
	[0.182]	[0.182]	[0.182]	[0.182]	[0.182]		
OECD	1.140***						
	[0.146]						
$OECD \times Democracy$ (political willingness)		1.516***					
		[0.171]					
$OECD \times Government effectiveness$			1.559***				
			[0.162]				
$OECD \times Regulatory environment$				1.680***			
				[0.158]			
$OECD \times Contract integrity$					1.530***		
					[0.158]		
Constant	-2.145***	-2.145***	-2.145***	-2.145***	-2.145^{***}		
	[0.175]	[0.175]	[0.175]	[0.175]	[0.175]		
Pseudo R-squared	0.185	0.191	0.192	0.191	0.187		
Log-Likelihood	-65.738	-63.993	-63.894	-65.217	-65.600		
Number of countries	133	131	131	133	133		

Note: e-GP and OECD are binary variables (0 or 1). Robust standard errors in brackets, ***;**,* indicate statistical significance at 1, 5 and 10% level, respectively. Source: Authors compilation.

cannot be viewed in isolation. The efficacy of one institution (such as an e-portals) is influenced by the efficacy of related institutions. For instance, e-participation in public policies and accountability increases citizen engagement and trust in government initiatives (Bisogno et al., 2022; Cucciniello & Nasi, 2014; Hall & Soskice, 2001; Pirannejad et al., 2019).

Second, we assign equal weights (25%) to the PPT-Index components. The results in Tables A.7-A10 confirm that transparency remains high in WTO-GPA countries. This is because the WTO-GPA requires 'sound' institutions which facilitate institutional reforms that provide, i. e. these provide access to procurement related information (awards, enotifications, e-quotations). However, this is not the case for the WTO- GPA Observers that are not mandated to comply with GPA commitments. A lower level of transparency in these countries can be attributed to unwillingness to reform (lack of political leadership) and poor institutional quality. Also, e-GP usage, the indirect effect of WTO-GPA and OECD membership (through institutional indicators, digital infrastructure and digital literacy skills)are considerably lower but yet remain highly significant. An explanation for these results is the weighting estimation scheme method which assigns the same importance to each dimension, and hence, underestimates the use of e-GP. Notwithstanding this, even with equal weight our main results hold.

Table 10

Transparency in Procurement, OECD membership and Information technology.

Variables \ Model specification	(I)	(II)	(III)	(IV)
	Dependent v	ariable: PPT-In	dex	
e-GP	2.549*** [0.181]	2.553*** [0.182]		
$OECD \times Digital$ infrastructure	2.221***		1.390***	
	[0.219]		[0.295]	
$OECD \times Digital literacy$		1.274*** [0.150]		0.851*** [0.177]
e-GP × Digital infrastructure			2.365***	
minuotracture			[0.432]	
e-GP \times Digital literacy				1.829*** [0.287]
Constant	-2.145*** [0.175]	-2.145*** [0.175]	-0.416*** [0.141]	-0.864*** [0.186]
Pseudo R-squared	0.192	0.187	0.165	0.169
Log-Likelihood	-65.199	-65.600	-67.352	-67.002
Number of countries	133	133	133	133

Note: e-GP and OECD are binary variables (0 or 1). Robust standard errors in brackets, ***,**,* indicate statistical significance at 1, 5 and 10% level, respectively.

Source: Authors compilation.

4.5.1. Sensitivity analysis

We conduct a sensitivity analysis with an enhanced Public Procurement Transparency Index (PPT_e -Index), this includes e-GP cycle; e-signature; e-bid challenge and e-public information The construct of bid challenge now includes two subcomponents namely, e-bid challenge mechanism and bid-challenge law. We differentiate between e-bid challenge and bid challenge law i.e. the possibility to perform an online recourse of Procurement Entity (PE) decisions and the legal requirement to provide information to unsuccessful bidders. We use the inclusion/exclusion procedure for the qualitative bid challenge law variable bid challenge law. A score of 1 is assigned if the country grants by law the possibility to challenge PE decisions legally. In cases where such information is missing i.e. not provided by governments or Public Procurement Law, a value of 0 is assigned. The inclusion of the sub-dimensions of bid-challenge informs whether excessive bureaucratic procedures disincentivize unsuccessful bidders to file bid-challenge.

We then re-run the PPT-Index with the expanded sub-components and confirm that the transparency index in public procurement is internally consistent and reliable. Fig. 9 displays the weights of Enhanced PPT-Index (PPT_e-I).

Table 11 reports the standardized loading factors and reliability of the index.

The factor loadings are above 0.4 which is widely accepted as a measure of stability (Stevens, 2002). An internal reliability analysis is performed to determine the consistency of the PPTe-I where a Cronbach's α of near here}0.7 indicates high reliability (Peterson, 1994). The Cronbach alpha for three of the four dimensions is above 0.7 with the exception of e-public information (0.57). Given that all items of the construct are binary (0 or 1), we employ the KR-20 test which is a special case of the Cronbach alpha (Crocker & Algina, 2008). The coefficient derived from the KR-20 test is near here} 0.8, which confirms good internal consistency. Finally, Fig. 10 displays the standard errors of the PPT-I and Enhanced PPT-I (PPT_e-I) for WTO-GPA, WTO-Observers and WTO non-member countries.

Results show that the PPTe-I has low dispersion (0.170) compared to the PPT-I (0.252). However, this is not the case for WTO-GPA parties (0.037 and 0.060) and WTO-Observers (0.077 and 0.164). The lower dispersion of the PPT-I is attributed to the tetrachoric analysis which reduces noise between the variables. For these reasons, we regard the PPT-I as our preferred measure of transparency in public procurement.

Table 11 Constructs, Items, and Standardized Factor Loadings.

Constructs	Items	Factor loadings
E-procurement cycle	(1) e-publishing	0.42
$(\alpha = 0.71)$	(2) contract award	0.56
	(3) online documents	0.70
E-signature functionalities	(1) e-signature advanced authentication	0.95
$(\alpha = 0.90)$	(2) e-signature all types	0.70
Bid-challenge	e-bid challenge	0.92
$(\alpha = 0.80)$	(2) bid challenge law	0.64
Access to public information	(1) e-reports	0.54
$(\alpha = 0.57)$	(2) e-statistics	0.55
	(3) legal gazette	0.52
	(4) disclosure public officials	0.41

[Notes: a) Convergent validity: 8/11 items (72.7%) have a correlation coefficient with the score of their own dimension greater than 0.400. Divergent validity: 4/11 items (36.4%) have a correlation coefficient with the score of their own dimension greater than those computed with other scores. b) The Comparative Fit Index (CFI) and Normal Fit Index (NFI) reported for the validation analysis are 0.937 and 0.912 respectively whereas the (standardized) root mean square residual is 0.068. CFI values higher than 0.90 and SRME lower than 0.08 indicate that the model improves the fit (Byrne, 1994; Fan et al., 1999; Schumacker & Lomax, 2004) Source: Authors compilation].



Fig. 9. Public Procurement Transparency Enhanced Index $(\mbox{PPT}_{e}\mbox{-}I)$ - weight of the components. Note: Authors compilation.



Fig. 10. Public Procurement Transparency (PPT—I) vs PPT-Enhanced Index (PPT_e-I).

Note: Authors compilation.

4.6. Limitations

The study has some limitations. First, the PPT-Index uses crosssectional data which means that the data allows running pooled models only. This restricts the ability to exploit the panel structure across different years and countries. Second, we use publicly available information such as GPPD, OECD Government at Glance reports and ERCAS which does not draw on surveys and expert opinions from the sample of countries analyzed. In line with Pirannejad et al. (2019), we acknowledge that a country's context (e.g. social factors and political context) could influence the weights of the Index and the release of procurement related information.

5. Policy implications

The governance of public procurement processes, which includes the use of e-GP tools complemented and the institutional framework, plays an important role in achieving and advancing the public policy objectives of higher transparency and accountability. Information on the existing levels of transparency increases participation in procurement and key starting points to assure sustainable and efficient public procurement systems. The findings of our results have two main policy implications. First, clear procurement regulations and a robust institutional framework provides access to laws and regulations, planned procurements, calls for tender, award announcements (domestic and foreign suppliers). Information relating to e-procurement systems (e.g. benchmarks, monitoring results, statistics, etc) promote fair and equitable treatment for potential suppliers, stakeholders and citizens. The findings suggest that the information within the procurement lifecycle (eGP usage; bid challenge mechanisms; use of e-signature for contracts and e-public information) has important implications for information usability, citizen participation and trust in government initiatives that in turn increase transparency. Along these lines, political and institutional leadership (e.g. joining the WTO-GPA) is decisive for the successful implementation of e-GP systems because it enables governments to lock in domestically contested economic reforms (Baccini & Urpelainen, 2015). Thus, the implementation of any e-GP system must be a change management process project and governments must ensure that political willingness is backed by ample information on the benefits of using e-GP to the procuring entities staff. Further, bidders must also be informed of the strategic importance of e-GP.

Second, a robust e-procurement framework system is important, but it is not sufficient to foster transparency. Strong institutions and e-GP portals enhance transparency at all levels of the procurement process, foster competition, and increase accountability of entities and staff. In addition, a coherent framework supporting the uptake of e-portals for

firms is vital to provide additional market access in government procurement. This is in line with the OECD (2016) that states access to public information is the cornerstone of an open and inclusive government to reduce corruption and increase trust among citizens and government. It is important to ensure that the staff in contracting entities and bidders have appropriate skills and knowledge of the processes and functionalities to use e-portals. Governments need to clearly define the vision for what is to be achieved through e-GP which must be backed by strong institutional leadership and political buy-in for reform. For the bidding entities and bidders to fully benefit, investment in digital infrastructure and skills for benefits is necessary. Policy interventions must promote public and private investment to improve the digital infrastructure that ensures connectivity and provides access to highspeed broadband internet and mobile networks. This means that boosting IT infrastructure requires availability, affordability, and high digital skills of citizens/stakeholders to use modern e-portals. In this manner, boosting and building digital skills awareness will require collaborative efforts by public, private and civic organisations. Policy makers must, therefore, focus on developing proactive policies that support the use of simple user-friendly e-portals through a robust digital infrastructure and digital skills will allow all users to leverage technology. These measures must be complemented by strong institutional quality to support competition and accountability in public procurement.

6. Conclusions

Three main findings emerge on transparency in public procurement. First, we develop a unique transparency index on public procurement. The novelty of the PPT-Index is its ability to amplify the flow of information at each stage: (e-GP cycle, mechanism of complains (bid challenge); support the adoption of e-signature of contracts; e-public information) of the entire procurement process which makes it a valuable tool for practitioners, citizens and policymakers in examining transparency, competition, integrity, accountability within the public procurement arena. Second. we find transparency levels are heterogenous across countries. The WTO-GPA signatories have higher transparency in public procurement compared to non-GPA members. The dissemination of procurement information depends to a large extent on the political decision (democracy) to join the WTO-GPA. The WTO-GPA mandates countries have robust institutions (namely government effectiveness, regulatory environment, contract integrity) which leads to higher transparency in procurement. Better access to procurement related information is in also linked with the political decision to join the WTO-GPA which creates an enabling business environment for citizens, stakeholders. Third, the level of IT infrastructure and digital skillsets are key drivers for higher transparency. A country's IT infrastructure (internet speed, connectivity) and digital skills-set of citizens/stakeholders in conjunction efficient functioning of e-GP tools supports higher access of bidders to procurement information. This supports eparticipation in government initiatives, increases accountability through the use of modern e-portals and promotes transparency in public procurement.

7. Directions for future research

Future studies should consider using longitudinal data on the PPT-Index variables (namely e-GP lifecycle, bid challenge recourse, esignature and e-public information) over time. This would mean analyzing longitudinal data. This could include for example contracts and/or surveys to assess whether bid challenge is likely to reduce corruption in public procurement. Another area of research should be examining whether the absence of checks and balances prevent bidders to challenge decisions on procurement entities. Thirdly, future researchers include social and political aspects of open government initiatives (O'Shaughnessy, Schiff, Varshney, Rozell, & Davenport, 2023)

S. Khorana et al.

and whether cyber security measures of e-portals improve contract security. Finally, the results are to be treated as associations rather than causal relations. It will also be interesting to explore whether specific initiative in procurement, such as sustainable green public procurement and gender-responsive public procurement are linked with the institutional framework of a country. Lastly, researchers could examine the link between the level of institutional framework and the degree of corruption in public procurement.

CRediT authorship contribution statement

Sangeeta Khorana: Writing - original draft, Methodology, Formal

Appendix A. Appendix

Table A1

List of countries.

analysis, Data curation, Conceptualization. **Santiago Caram:** Writing – original draft, Validation, Formal analysis, Data curation. **Nripendra P. Rana:** Writing – review & editing, Supervision, Methodology, Formal analysis.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Panel A: GPA members (4	3)				
Armenia	Czech, Rep.	Iceland	Luxembourg	Portugal	Ukraine
Australia	Denmark	Ireland	Malta	Romania	United Kingdom
Austria	Estonia	Israel	Moldova	Singapore	United States
Belgium	Finland	Italy	Montenegro	Slovak, Rep).
Bulgaria	France	Japan	Netherlands	Slovenia	
Canada	Germany	Korea, Rep.	New Zealand	Spain	
Croatia	Greece	Latvia	Norway	Sweden	
Cyprus	Hungary	Lithuania	Poland	Switzerland	1
-5F					
Panel B: Observer status (31)				
Albania	Colombia	India	Mongolia		Russia Vietnam
Argentina	Costa Rica	Indonesia	North Macedonia		Saudi Arabia
Brazil	Côte d'Ivoire	Jordan	Pakistan		Sri Lanka
Cameroon	Dominican, Rep.	Kazakhstan	Panama		Tajikistan
Chile	Ecuador	Kyrgyz, Rep.	Paraguay		Thailand
China	Georgia	Malaysia	Philippines		Turkey
Panel C: WTO-non-membe	ers (59)				
Algeria	Burkina Faso	Guatemala	Malawi	Peru	Trinidad & Tobago
Angola	Burundi	Guyana	Mali	Qatar	Tunisia
Azerbaijan	Cambodia	Haiti	Mexico	Rwanda	Uganda
Bangladesh	Chad	Honduras	Morocco	Senegal	Uruguay
Barbados	Congo DRC	Iran	Mozambique	Serbia	Uzbekistan
Belize	Egypt, Arab Rep.	Jamaica	Myanmar	Sierra Leone	Venezuela
Benin	El Salvador	Kenya	Namibia	South Africa	Yemen
Bolivia	Ethiopia	Lebanon	Nepal	St. Lucia	Zambia
Bosnia & Herzg.	Ghana	Lesotho	Nicaragua	Tanzania	Zimbabwe
Botswana	Grenada	Liberia	Nigeria	Timor-Leste	
Panel D: Total number of	countries excluded in multi-stage	e sampling due to non-availability	of information in GPPD databa	se (85)	
Afghanistan	Congo, Rep.	Hong Kong	Micronesia		South Sudan
American Samoa	Cuba	Iraq	Monaco		St. Kitts & Nevis
Andorra	Curaçao	Isle of Man	Nauru		St. Martin (French part)
Antigua and Barbuda	Djibouti	Kingdom of Eswatini	New Caledonia		St. Vincent & the Grenadines
Aruba	Dominica	Kiribati	Niger		Sudan
Bahamas, The	Equatorial Guinea	Korea, Dem. Rep.	Northern Mariana	Isl.	Suriname
Bahrain	Eritrea	Kosovo	Oman		Syrian Arab Rep.
Belarus	Faroe Islands	Kuwait	Palau		Taiwan
Bermuda	Fiji	Lao PDR	Papua New Guine	а	Togo
Bhutan	French Polynesia	Libya	Puerto Rico		Tonga
British Virgin Islands	Gabon	Liechtenstein	Samoa		Turkmenistan
Brunei Darussalam	Gambia, The	Macao, China	San Marino		Turks & Caicos Islands
Cabo Verde	Gibraltar	Madagascar	São Tomé & Princ	ripe	Tuvalu
Cayman Islands	Greenland	Maldives	Seychelles		United Arab Emirates
Central African Rep.	Guam	Marshall Islands	Sint Maarten		Vanuatu
Channel Islands	Guinea	Mauritania	Solomon Islands		Virgin Islands (U.S.)
Comoros	Guinea-Bissau	Mauritius	Somalia		West Bank and Gaza

Source: Authors compilation.

Table A2

_

Measurements Results Public Procurement Transparency (PPT-I).

Panel A: Factor Analysis		
	Loading factor1	Uniqueness
Legal gazette	0.648	0.581
Public official declarations	0.692	0.520
E-procurement reports & statistics	0.610	0.628
Panel B: Internal reliability	Average interitem correlation	Cronbach's alpha
e-GP cycle	0.552	0.787

Test scale	0.432	
E-public information	0.357	
E-signature	0.416	
Bid challenge	0.402	

Source: Authors compilation.

Table A3

Database, variables and sources.

Dataset	Variable	Source
World Trade Organisation (WTO)	WTO-GPA countries	https://www.wto.org/english/tratop_e/gproc_e/mem
	WTO-Observer countries	obs_e.htm
Global Public Procurement Database (GPPD)	e-notification; online documents; awards; e-signature advanced	https://www.globalpublicprocurementdata.org
	authentication; e-signature other methods; e-reports; e-statistics	
European Research Centre for Anticorruption	legal gazette; disclosure public officials	https://zenodo.org/records/7225627
and State-building (ERCAS)		www.corruptionrisk.org/transparency.
(OECD, 2017, OECD, 2019a; OECD, 2019b;	e-notification; online documents; awards; e-signature advanced	https://doi.org/10.1787/gov_glance-2017-en
OECD, 2019c, OECD, 2021)	authentication; e-signature other methods; e-reports; e-statistics	https://doi.org/10.1787/8ccf5c38-en.
		https://doi.org/10.1787/1c258f55-en
		https://www.oecd.org/about/members-and-partners/
World Governance Indicators (WGI)	Democracy (eiu21va1) in 2021	http://www.govindicators.org
	Government effectiveness (eiu21ge1) in 2021	
	Regulatory environment (wmo21rq1) in 2021	
	Contract integrity (wmo21cc1) in 2021	
World Development Indicators (WDI)	Fixed broadband subscriptions (per 100 people), (IT.NET.BBND-P2) in	http://data.worldbank.org/indicator/IT.NET.BBND.P2
	2021	http://data.worldbank.org/indicator/IT.CEL.SETS.P2
	Mobile cellular subscriptions (per 100 people), (IT.CEL.SETS-P2) in	http://data.worldbank.org/indicator/EG.ELC.ACCS.ZS
	2021	
	Access to electricity (% of population), (EG.ELC.ACCS.ZS) in 2021	
International Telecommunication Union (ITU)	International bandwidth per Internet user (bit/s) in 2021	https://datahub.itu.int/data/
Digital Development Compass Dataset	Digital literacy skills in 2022	http://www.digitaldevelopmentcompass.org
		https://github.com/undp/digital-development-
		compass
Bid-challenge	e-bid challenge	Author's based on manual search from e-portals
	bid challenge law	GPPD (Public Procurement Law),
		https://www.globalpublicprocurementdata.org
		https://bpp.worldbank.org/content/dam/sites/data/b
		pp/documents/BID-2023-PPP-Dataset.xlsx

Source: Authors compilation.

Table A4

Country	PPT-I (overall score)
Albania	0.748
Algeria	0.037
Angola	0.563
Argentina	0.645
Armenia	1
Australia	0.955
Austria	0.955
Azerbaijan	0.748
Bangladesh	0.748
Barbados	0.000
Belgium	0.955
Belize	0.000
Benin	0.563
	(continued on next page)

Public Procurement Transparency (PPT-I) scores of the _

0.669 0.681 0.625 0.752

Table A4 (continued)

Country	PPT-I (overall score)
Bolivia	0.560
Bosnia and Herzegovina	0.748
Botswana	0.563
Brazil	0.600
Bulgaria Burkina Faco	1
Burundi	0.600
Cambodia	0.037
Cameroon	0.711
Canada	0.955
Chile	0.000
China	0.600
Colombia	0.605
Congo (DRC)	0.563
Costa Rica Côte d'Ivoire	0.560
Croatia	1
Cyprus	0.917
Czech Republic	0.955
Denmark	0.955
Dominican Republic	0.608
Egypt, Arab Rep	0.600
El Salvador	0.748
Estonia	1
Ethiopia Finland	0.711
France	1
Georgia	0.793
Germany	0.955
Ghana	0.563
Greece Grenada	1 0.523
Guatemala	0.600
Guyana	0.563
Haiti	0.563
Hungary	0.600
Iceland	0.955
India	0.748
Indonesia	0.748
Iran Ireland	0.560
Israel	1
Italy	1
Jamaica	0.563
Japan Jordan	0.955
Kazakhstan	0.711
Kenya	0.600
Korea, Rep.	1
Kyrgyz Republic	0.748
Lebanon	0.000
Lesotho	0.523
Liberia	0.563
Lithuania	1
Malawi	0.563
Malaysia	0.600
Mali	0.600
Malta	0.955
Moldova	1
Mongolia	0.600
Montenegro	0.852
Morocco	0.748
wozambique Myanmar	0.503
Namibia	0.000
Nepal	0.807
Netherlands	0.955
New Zealand Nicaragua	0.852
Nigeria	0.563
-	(continued on next page)

Country	PPT-I (overall score)
North Macedonia	0.748
Norway	0.955
Pakistan	0.708
Panama	0.600
Paraguay	0.645
Peru	0.793
Philippines	0.600
Poland	0.955
Portugal	1
Qatar	0.748
Romania	1
Russian Federation	0.748
Rwanda	0.748
Saudi Arabia	0.600
Senegal	0.600
Serbia	0.600
Sierra Leone	0.000
Singapore	0.914
Slovak Republic	1
Slovenia	0.955
South Africa	0.563
Spain	1
Sri Lanka	0.748
St. Lucia	0.523
Sweden	0.955
Switzerland	1
Tajikistan	0.600
Tanzania	0.563
Thailand	0.793
Timor-Leste	0.600
Trinidad and Tobago	0.600
Tunisia	0.600
Turkey	0.748
Uganda	0.563
Ukraine	1
United Kingdom	1
United States of America	0.955
Uruguay	0.600
Uzbekistan	0.600
Venezuela	0.560
Vietnam	0.748
Yemen	0.563
Zambia	0.711
Zimbabwe	0.523

Source: Authors compilation.

Table A5

Pairwise correlations.

Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
(1) PPT-Index	1									
(2) e-GP	0.653***	1								
(3) WTO-GPA	0.725***	0.166*	1							
(4) WTO-GPA non-member	-0.627***	-0.271***	-0.613***	1						
(5) Democracy (political willingness)	0.538***	0.261***	0.711***	-0.528***	1					
(6) Institutional effectiveness	0.622***	0.265***	0.694***	-0.632^{***}	0.820***	1				
(7) Regulatory environment	0.560***	0.238***	0.637***	-0.551***	0.735***	0.811***	1			
(8) Contract integrity	0.485***	0.149*	0.690***	-0.499***	0.803***	0.825***	0.778***	1		
(9) Digital infrastructure	0.632***	0.248***	0.736***	-0.649***	0.709***	0.779***	0.755***	0.684***	1	
(10) Digital literacy	0.595***	0.289***	0.636***	-0.677***	0.659***	0.725***	0.631***	0.576***	0.844***	1

Note: e-GP, WTO-GPA and WTO-GPA non-member are binary variables (0 or 1). ***, **, * indicate statistical significance at 1, 5 and 10% level, respectively. Source: Authors compilation.

Table A6	
Multicollinearity analysis.	

Variables	VIF	Tolerance
PPT-Index	7.070	0.142
e-GP	3.130	0.320
WTO-GPA	7.420	0.135
WTO-GPA Observer	1.680	0.594
	(co	ntinued on next page)

Table A6 (continued)
------------	-----------	---

Variables	VIF	Tolerance
OECD	3.310	0.302
Democracy (political willingness)	4.500	0.222
Institutional effectiveness	6.590	0.152
Regulatory environment	4.060	0.246
Contract integrity	4.730	0.211
Digital infrastructure	5.710	0.175
Digital literacy	4.300	0.233
Mean	4.773	0.248

Notes: a. e-GP, WTO-GPA, WTO-GPA Observer and OECD are binary variables (0 or 1).

b. VIF values greater than 10 indicate reasons for concern due to collinearity among variables. Tolerance values less than 0.1 indicate collinearity among variables. From our (centred) values we can infer that the majority of variables do not suffer from severe collinearity issues. Another useful rule of thumb to check for the presence of multicollinearity is the condition number of the eigenvalues below 15. In our case, the condition (not reported) is 10.089 which falls within that rule. Source: Authors compilation.

Table A7

Transparency in Procurement, WTO Membership and Institutional quality.

Variables \setminus Model specification	(I)	(II)	(III)	(IV)	(V)
	Dependent variable: PPT-Index (equal weights)				
e-GP	1.631*** [0.205]	1.696*** [0.205]	1.727*** [0.205]	1.654*** [0.205]	1.700*** [0.205]
WTO-GPA	1.765***				
	[0.105]				
WTO-GPA Observer	0.307***				
WTO-GPA \times Democracy (political willingness)	[0.080]	2.195***			
		[0.156]			
WTO-GPA Observer \times Democracy (political willingness)		0.454**			
		[0.179]	0 101+++		
WIO-GPA × Government effectiveness			2.191***		
WTO-GPA Observer \times Government effectiveness			0.469**		
			[0.190]		
WTO-GPA \times Regulatory environment				2.304***	
WTO CDA Observer × Regulatory environment				[0.163]	
W10-GFA Observer × Regulatory environment				[0.152]	
WTO-GPA \times Contract integrity					2.179***
					[0.185]
WTO-GPA Observer \times Contract integrity					0.527**
Constant	-1.845***	-1.845***	-1.845***	-1.845***	[0.199] -1.845***
Constant	[0.196]	[0.196]	[0.196]	[0.196]	[0.196]
Pseudo R-squared	0.249	0.238	0.226	0.240	0.227
Log-Likelihood	-67.509	-67.262	-68.271	-68.345	-69.462
Number of countries	133	131	131	133	133

Note: e-GP, WTO-GPA and WTO-GPA Observers are binary variables (0 or 1). Robust standard errors in brackets, ***;**,* indicate statistical significance at 1, 5 and 10% level respectively.

Source: Authors compilation.

Table A8

Transparency in Procurement, WTO Membership and Information technology.

Variables \setminus Model specification	(I)	(II)	(III)	(IV)	
	Dependent variable: PPT-Index (equal weights)				
e-GP	1.667***	1.638***			
	[0.205]	[0.205]			
WTO-GPA \times Digital Infrastructure	2.943***		2.455***		
	[0.245]		[0.368]		
WTO-GPA Observer \times Digital infrastructure	0.696**		0.527*		
	[0.229]		[0.311]		
WTO-GPA \times Digital literacy		1.931***		1.633***	
		[0.119]		[0.181]	
WTO-GPA Observer \times Digital literacy		0.388***		0.214	
		[0.110]		[0.155]	

(continued on next page)

Table A8 (continued)

Variables \ Model specification	(I)	(II)	(III)	(IV)	
	Dependent variable: PPT-Index (equal weights)				
e-GP ⁺ × Digital infrastructure			1.184**		
			[0.504]		
e-GP \times Digital literacy				1.086**	
				[0.312]	
Constant	-1.845^{***}	-1.845^{***}	-0.571***	-0.920***	
	[0.196]	[0.196]	[0.135]	[0.173]	
Pseudo R-squared	0.236	0.247	0.213	0.231	
Log-Likelihood	-68.633	-67.712	-70.762	-69.086	
Number of countries	133	133	133	133	

Note: e-GP, WTO-GPA and WTO-GPA Observers are binary variables (0 or 1). Robust standard errors in brackets, ***;**,* indicate statistical significance at 1, 5 and 10% level respectively.

Source: Authors compilation.

Table A9

Transparency in Procurement, OECD Membership and Institutional quality.

Variables/Model specification	(I)	(II)	(III)	(IV)	(V)
	Dependent variable: PPT-Index (equal weights)				
e-GP	1.859***	1.875***	1.876***	1.854***	1.868***
	[0.204]	[0.204]	[0.204]	[0.204]	[0.204]
OECD	1.260***				
	[0.149]				
OECD \times Democracy (political willingness)		1.640***			
		[0.174]			
$OECD \times Government effectiveness$			1.682***		
			[0.164]		
$OECD \times Regulatory environment$				1.826***	
				[0.162]	
$OECD \times Contract integrity$					1.654***
					[0.161]
Constant	-1.845***	-1.845***	-1.845***	-1.845^{***}	-1.845^{***}
	[0.196]	[0.196]	[0.196]	[0.196]	[0.196]
Pseudo R-squared	0.171	0.175	0.177	0.179	0.172
Log-Likelihood	-74.467	-72.794	-72.637	-73.801	-74.442
Number of countries	133	131	131	133	133

Note: e-GP and OECD are binary variables (0 or 1). Robust standard errors in brackets, ***;**,* indicate statistical significance at 1, 5 and 10% level respectively. Source: Authors compilation.

Table A10

Transparency in Procurement, OECD Membership and Information technology.

Variables \ Model specification	(I)	(II)	(III)	(IV)	
	Dependent variable: PPT-Index (equal weights)				
e-GP	1.853***	1.857***			
	[0.204]	[0.204]			
$OECD \times Digital infrastructure$	2.418***		1.543***		
	[0.224]		[0.296]		
$OECD \times Digital literacy$		1.402***		0.983***	
		[0.153]		[0.180]	
$eGP \times Digital infrastructure$			2.376***		
			[0.369]		
eGP \times Digital literacy				1.759***	
				[0.258]	
Constant	-1.845^{***}	-1.845***	-0.802***	-1.203^{***}	
	[0.196]	[0.196]	[0.113]	[0.157]	
Pseudo R-squared	0.179	0.173	0.188	0.181	
Log-Likelihood	-73.760	-74.294	-72.983	-73.615	
Number of countries	133	133	133	133	

Note: e-GP and OECD are binary variables (0 or 1). Robust standard errors in brackets, ***;**,* indicate statistical significance at 1, 5 and 10% level respectively.

Source: Authors compilation.

References

Ackerman, J. M., & Sandoval-Ballesteros, I. E. (2006). The global explosion of freedom of information laws. Administrative Law Review, 58(85), 85–130.

Adam, I., Fazekas, M., & Tóth, B. (2020). Measuring the benefits of open contracting. Government Transparency Institute.

Aladwani, A. M. (2016). Corruption as a source of e-government projects failure in developing countries: A theoretical exposition. *International Journal of Information Management*, 36(1), 105–112. https://doi.org/10.1016/j.ijinfomgt.2015.10.005

- Allee, T. L., & Scalera, J. E. (2012). The divergent effects of joining international organizations: Trade gains and the rigors of WTO accession. *International Organization*, 66(2), 243–276. https://doi.org/10.1017/S0020818312000082
- Alt, J. E., & Lowry, R. C. (1994). Divided government, fiscal institutions, and budget deficits: Evidence from the states. *American Political Science Review*, 88(4), 811–828. https://doi.org/10.2307/2082709
- Anderson, M. L. (2008). Multiple inference and gender differences in the effects of early intervention: A reevaluation of the abecedarian, perry preschool, and early training projects. Journal of the American Statistical Association, 103, 1481–1495. https://doi. org/10.2307/2118402
- Armstrong, E. (2005). Integrity, transparency and accountability in public administration: Recent trends, regional and international developments and emerging issues. New York: United Nations. Online access https://demolive.insightsonindia.com/wp-content/up loads/2013/09/integrity-transparency-un.pdf.
- Baccini, L., & Urpelainen, J. (2015). Cutting the gordian knot of economic reform: When and how international institutions help. USA: Oxford University Press.
- Bandura, R. (2008). A survey of composite indices measuring country performance: 2008 update (technical report). In Office of development studies. United Nations Development. Online access: https://www.undp.org/content/dam/undp/library/c orporate/development%20studies/indices_2008_bandura.pdf.
- Bannister, F. (2007). The curse of the benchmark: An assessment of the validity and value of e-government comparisons. *International Review of Administrative Sciences*, 73(2), 171–188. https://doi.org/10.1177/0020852307077959
- Barcevičius, E., Čibaitė, G., Čodagnone, C., Gineikytė, V., Klimavičiūtė, L., Liva, G., ... Vanini, I. (2019). Exploring digital government transformation in the EU: Analysis of the state of the art and review of literature. EUR 29987 EN. Luxembourg: Publications Office of the European Union. Online access: https://core.ac.uk/downloa d/pdf/268886171.pdf.
- Bauhr, M., & Grimes, M. (2014). Indignation or resignation: The implications of transparency for societal accountability. *Governance*, 27(2), 291–320. https://doi. org/10.1111/gove.12033
- Beblavý, M., Sičáková-Beblavá, E., & Bačová, B. (2022). A conceptual framework for understanding and measuring the transparency of public policy. *Sociologia*, 54(2), 10.31577/sociologia.2022.54.2.4.
- Becker, J. (2018). Systems and e-procurement-improving access and transparency of public procurement. Brussels, Belgium. https://doi.org/10.2861/828878.
- Bendoly, E., & Schoenherr, T. (2005). ERP system and implementation-process benefits: Implications for b2b e-procurement. International Journal of Operations & Production Management. https://doi.org/10.1108/01443570510585516
- Bertot, J. C., Jaeger, P. T., & Grimes, J. M. (2010). Using ICTs to create a culture of transparency: E-government and social media as openness and anti-corruption tools for societies. *Government Information Quarterly*, 27(3), 264–271. https://doi.org/ 10.1016/j.giq.2010.03.001
- Beverelli, C., Keck, A., Larch, M., & Yotov, Y. (2018). Institutions, trade and development: A quantitative analysis. CeSifo Working Papers, 6920 February.
- Bhuiyan, S. H. (2011). Modernizing Bangladesh public administration through egovernance: Benefits and challenges. *Government Information Quarterly*, 28(1), 54–65. https://doi.org/10.1016/j.giq.2010.04.006
- Bisogno, M., Cuadrado-Ballesteros, B., & Santis, S. (2022). Do e-government initiatives and e-participation affect the level of budget transparency? *International Public Management Journal*, 25(3), 365–391. https://doi.org/10.1080/ 10967494.2022.2042437
- Bosio, E., Djankov, S., Glaeser, E., & Shleifer, A. (2022). Public procurement in law and practice. American Economic Review, 112(4), 1091–1117.
- Brunswicker, S., Pujol Priego, L., & Almirall, E. (2019). Transparency in policy making: A complexity view. Government Information Quarterly, 36(3), 571–591. https://doi.org/ 10.1016/j.giq.2019.05.005
- Bun, M. J., & Harrison, T. D. (2019). OLS and IV estimation of regression models including endogenous interaction terms. *Econometric Reviews*, 38(7), 814–827. https://doi.org/10.1080/07474938.2018.1427486
- Byrne, B. M. (1994). Structural equation modeling with EQS and EQS/windows: Basic concepts, applications, and programming. Sage.
- Carothers, T., & Gramont, D. (2011). Aiding governance in developing countries: Progress amid uncertainties. In *Carnegie endowment for international peace*. Washington, D. C., USA.
- Chau, P. Y., & Tam, K. Y. (1997). Factors affecting the adoption of open systems: An exploratory study. *MIS Quarterly*, 1–24. https://doi.org/10.2307/249740
- Chen, C., & Neshkova, M. I. (2020). The effect of fiscal transparency on corruption: A panel cross-country analysis. *Public Administration*, 98(1), 226–243. https://doi.org/ 10.1111/padm.12620
- Chen, Y., Bretschneider, S., Stritch, J. M., Darnall, N., & Hsueh, L. (2022). E-procurement system adoption in local governments: The role of procurement complexity and organizational structure. *Public Management Review*, 24(6), 903–925. https://doi. org/10.1080/14719037.2021.1874497
- Cho, Y. H., & Choi, B. D. (2005). e-Government to combat corruption: the case of Seoul Metropolitan Government. *International Journal of Public Administration*, 27(10), 719–735. https://doi.org/10.1081/PAD-200029114
- Choudhury, S. (2019). WTO membership and corruption. European Journal of Political Economy, 60, Article 101806. https://doi.org/10.1016/j.ejpoleco.2019.07.004
- Ciborra, C. (2005). Interpreting e-government and development: Efficiency, transparency or governance at a distance? *Information Technology & People*, 18(3), 260–279. https://doi.org/10.1108/09593840510615879
- Cinar, E., Trott, P., & Simms, C. (2019). A systematic review of barriers to public sector innovation process. *Public Management Review*, 21(2), 264–290. https://doi.org/ 10.1080/14719037.2018.1473477

- Cinar, E., Trott, P., & Simms, C. (2021). An international exploration of barriers and tactics in the public sector innovation process. *Public Management Review*, 23(3), 326–353. https://doi.org/10.1080/14719037.2019.1668470
- Citro, F., Cuadrado-Ballesteros, B., & Bisogno, M. (2021). Explaining budget transparency through political factors. *International Review of Administrative Sciences*, 87(1), 115–134. https://doi.org/10.1177/0020852319847511
- Clarkson, D. B., & Jennrich, R. I. (1988). Quartic rotation criteria and algorithms. Psychometrika, 53(2), 251–259. https://doi.org/10.1007/BF02294136
- Correa, A. S., de Souza, R. M., & da Silva, F. S. C. (2019). Towards an automated method to assess data portals in the deep web. *Government Information Quarterly*, 36(3), 412–426. https://doi.org/10.1016/j.giq.2019.03.004
- Crocker, L., & Algina, J. (2008). Introduction to classical and modern test theory. New York, NY: Holt.
- Croom, S. R. (2000). The impact of web-based procurement on the management of operating resources supply. *Journal of Supply Chain Management*, 36(4), 4–13. https://doi.org/10.1111/j.1745-493X.2000.tb00065.x
- Cucciniello, M., & Nasi, G. (2014). Transparency for trust in government: How effective is formal transparency? *International Journal of Public Administration*, 37(13), 911–921. https://doi.org/10.1080/01900692.2014.949754
- Djankov, S., Ghossein, T. M., Islam, A., & Saliola, F. (2017). Public procurement regulation and road quality. In, 8234. World bank policy research working paper.
- Drabek, Z., & Bacchetta, M. (2004). Tracing the effects of WTO accession on policymaking in sovereign states: Preliminary lessons from the recent experience of transition countries. *The World Economy*, 27(7), 1083–1125. https://doi.org/ 10.1111/j.0378-5920.2004.00639.x
- Duguay, R., Rauter, T., & Samuels, D. (2023). The impact of open data on public procurement. *Journal of Accounting Research*, 61(4), 1159–1224. https://doi.org/ 10.1111/1475-679X.12479
- Elbahnasawy, N. G. (2014). E-government, internet adoption, and corruption: An empirical investigation. World Development, 57, 114–126. https://doi.org/10.1016/j. worlddev.2013.12.005
- Eppel, E. (2012). The application of a complexity analytical Lens to explain and understand public policy processes. In L. Gerrits, & P. Marks (Eds.), COMPACT 1: Public administration in complexity (pp. 174–192). Litchfield Park, AZ: ISCE.
- Essig, M., & Arnold, U. (2001). Electronic procurement in supply chain management: An information economics-based analysis of electronic markets. *Journal of Supply Chain Management*, 37(3), 43–49. https://doi.org/10.1111/j.1745-493X.2001.tb00112.x
- Evans, D., & Yen, D. C. (2005). E-government: An analysis for implementation: Framework for understanding cultural and social impact. *Government Information Quarterly*, 22(3), 354–373. https://doi.org/10.1016/j.giq.2005.05.007
- Evenett, S. J., & Hoekman, B. M. (2003). Transparency in government procurement: What can we expect from international trade agreements. In S. Arrowsmith, & M. Trybus (Eds.), *Public procurement: The continuing revolution* (pp. 269–282). MA: Kluwer Law International.
- Evenett, S. J., & Hoekman, B. M. (2005). International cooperation and the reform of public procurement policies. In , 3720. World bank policy research working paper.
- Fan, Q., Kuper, P., Choi, Y. H., & Choi, S. J. (2021). Does ICT development curb firms' perceived corruption pressure? The contingent impact of institutional qualities and competitive conditions. *Journal of Business Research*, 135, 496–507. https://doi.org/ 10.1016/j.jbusres.2021.06.062
- Fan, X., Thompson, B., & Wang, L. (1999). Effects of sample size, estimation methods, and model specification on structural equation modeling fit indexes. *Structural Equation Modeling: A Multidisciplinary Journal, 6*(1), 56–83. https://doi.org/10.1080/ 10705519909540119

Fatas, A., & Mihov, I. (2013). Policy volatility, institutions, and economic growth. Review of Economics and Statistics, 95(2), 362–376. https://doi.org/10.1162/REST_a_00265

- Fazekas, M., Lukács, P. A., & Tóth, I. J. (2015). The political economy of grand corruption in public procurement in the construction sector of Hungary. *Government Favouritism in Europe the Anticorruption Report*, *3*, 53–68.
- Fish, M. S., Michel, K., & Lindberg, S. (2015). Legislative powers and executive corruption. Gothenburg: The Varieties of Democracy Institute.
- Fukuyama, F. (2013). What is governance? Governance: An international journal of policy. Administration, and Institutions, 26(3), 347–368. https://doi.org/10.1111/ gove.12035
- Ghossein, T., Islam, A. M., & Saliola, F. (2018). Public procurement and the private business sector. In , 8575. World Bank Policy Research Working Paper.
- Global Public Procurement Database. (2020). The World Bank Group. https://www.gl obalpublicprocurementdata.org.
- Grigorescu, A. (2003). International organizations and government transparency: Linking the international development realms. *International Studies Quarterly*, 47, 643–667. https://doi.org/10.1046/j.0020-8833.2003.04704003.x
- Gunasekaran, A., & Ngai, E. W. (2008). Adoption of e-procurement in Hong Kong: An empirical research. *International Journal of Production Economics*, 113(1), 159–175. https://doi.org/10.1016/j.ijpe.2007.04.012
- Gupta, M., & Narain, R. (2012). Investigation into barriers to adoption of e-procurement and measures of performance. *International Journal of Procurement Management*, 5(5), 567–607. https://doi.org/10.1504/IJPM.2012.048877
- Haddara, M., & Elragal, A. (2013). ERP adoption cost factors identification and classification: A study in SMEs. International Journal of Information Systems and Project Management, 1(2), 5–21. https://doi.org/10.12821/ijispm010201
- Hall, P. A., & Soskice, D. W. (Eds.). (2001). Varieties of capitalism: The institutional foundations of comparative advantage. Oxford: Oxford University Press.
- Hardy, C. A., & Williams, S. P. (2008). E-government policy and practice: A theoretical and empirical exploration of public e-procurement. *Government Information Quarterly*, 25(2), 155–180. https://doi.org/10.1016/j.giq.2007.02.003

Harrison, T. M., & Sayogo, D. S. (2014). Transparency, participation, and accountability practices in open government: A comparative study. *Government Information Quarterly*, 31(4), 513–525. https://doi.org/10.1016/j.giq.2014.08.002

Heeks, R. (2005). E-government as a carrier of context. Journal of Public Policy, 25(1), 51–74. https://doi.org/10.1017/S0143814X05000206

- Heilbrunn, J. R. (2004). Anti-corruption commissions: Panacea or real medicine to fight corruption. Washington D.C.: World Bank Institute.
- Henriksen, H. Z., & Mahnke, V. (2005). E-procurement adoption in the danish public sector: The influence of economic and political rationality. *Scandinavian Journal of Information Systems*, 17(2), 2.
- Hujran, O., Al-Debei, M. M., Al-Adwan, A. S., Alarabiat, A., & Altarawneh, N. (2023). Examining the antecedents and outcomes of smart government usage: An integrated model. *Government Information Quarterly*, 40(1), Article 101783. https://doi.org/ 10.1016/j.giq.2022.101783

International Telecommunication Union (ITU). (2011). Measuring the information society. Annual report of International Telecommunication Union. Geneva: ITU.

- Islam, R. (2003). Do more transparent governments govern better? Policy research working paper 3077. Online access: https://documents1.worldbank.org/curated/e n/568401468741328131/pdf/multi0page.pdf.
- Islam, R. (2006). Does more transparency go along with better governance? Economics and Politics, 18(2), 121–167. https://doi.org/10.1111/j.1468-0343.2006.00166.x

Janssen, M., Hartog, M., Matheus, R., Yi Ding, A., & Kuk, G. (2022). Will algorithms blind people? The effect of explainable AI and decision-makers' experience on AIsupported decision-making in government. *Social Science Computer Review*, 40(2), 478–493. https://doi.org/10.1177/0894439320980118

- Jiménez, A., Hanoteau, J., & Barkemeyer, R. (2022). E-procurement and firm corruption to secure public contracts: The moderating role of governance institutions and supranational support. *Journal of Business Research*, 149, 640–650. https://doi.org/ 10.1016/j.jbusres.2022.05.070
- Kabanov, Y. (2022). Refining the UN E-participation index: Introducing the deliberative assessment using the varieties of democracy data. *Government Information Quarterly*, 39(1), Article 101656. https://doi.org/10.1016/j.giq.2021.101656

Kameshwaran, S., Narahari, Y., Rosa, C. H., Kulkarni, D. M., & Tew, J. D. (2007). Multiattribute electronic procurement using goal programming. *European Journal of Operational Research*, 179(2), 518–536. https://doi.org/10.1016/j.ejor.2006.01.010 Kaufmann, D., & Kraay, A. (2023). Worldwide governance indicators. Update

http://www.govindicators.org.

Kaufmann, D., Kraay, A., & Mastruzzi, M. (2006). Governance matters V: Aggregate and individual governance indicators for 1996–2005. 4012. World Bank Publications.

Kaufmann, D., Kraay, A., & Mastruzzi, M. (2007). Growth and governance: A reply. *The Journal of Politics*, 69(2), 555–562. https://doi.org/10.1111/j.1468-2508.2007.00550.x

Kaufmann, D., Kraay, A., & Mastruzzi, M. (2010). The worldwide governance indicators: Methodology and analytical issues. In , 5430. World bank policy research working paper.

Kearns, I. (2004). Public value and e-government. Institute for Public Policy Research. Online access https://citeseerx.ist.psu.edu/document?repid=rep1&type=pdf&doi =1959beb80056dbb823ae035303361b6285e4a480.

Khorana, S., Ferguson-Boucher, K., & Kerr, W. A. (2015). Governance issues in the EU's e-procurement framework. JCMS: Journal of Common Market Studies, 53(2), 292–310. https://doi.org/10.1111/jcms.12179

- Khorana, S., & Kerr, W. A. (2021). Electronic government procurement in the EU-Vietnam free trade agreement: An opportunity for increased transparency and accountability? *Journal of World Trade*, 55(1). https://doi.org/10.54648/ trad2021005
- Khorana, S., Mishra, N., & Kerr, W. A. (2014). Transforming Vietnam: A quest for improved efficiency and transparency in central government procurement. *Policy & Politics*, 42(1), 125–145. https://doi.org/10.1332/030557312X655611
- Kim, C. K. (2007). A cross-national analysis of global E-government. Public Organization Review, 7, 317–329. https://doi.org/10.1007/s11115-007-0040-5
- Kim, S., Kim, H. J., & Lee, H. (2009). An institutional analysis of an e-government system for anti-corruption: The case of open. *Government Information Quarterly*, 26(1), 42–50. https://doi.org/10.1016/j.giq.2008.09.002
- Kim, S., & Lee, J. (2012). E-participation, transparency, and trust in local government. *Public Administration Review*, 72(6), 819–828. https://doi.org/10.1111/j.1540-6210.2012.02593.x
- Klabi, H., Mellouli, S., & Rekik, M. (2018). A reputation based electronic government procurement model. *Government Information Quarterly*, 35(4), S43–S53. https://doi. org/10.1016/j.giq.2016.01.001
- Knack, S., Biletska, N., & Kacker, K. (2019). Deterring kickbacks and encouraging entry in public procurement markets: Evidence from firm surveys in 90 developing countries. *The World Bank Economic Review*, 33(2), 287–309. https://doi.org/ 10.1093/wber/lhy016

Kolenikov, S., & Angeles, G. (2004). The use of discrete data in PCA: Theory, simulations, and applications to socioeconomic indices. In , 20. Chapel hill: Carolina population center (pp. 1–59). University of North Carolina.

Kolenikov, S., & Angeles, G. (2009). Socioeconomic status measurement with discrete proxy variables: Is principal component analysis a reliable answer? *Review of Income* and Wealth, 55(1), 128–165. https://doi.org/10.1111/j.1475-4991.2008.00309.x

- Kono, D. Y., & Rickard, S. J. (2014). Buying national: Democracy, public procurement, and international trade. *International Interactions*, 40(5), 657–682. https://doi.org/ 10.1080/03050629.2014.899220
- Križić, I. (2021). Regulating public procurement in Brazil, India, and China: Toward the regulatory-developmental state. *Regulation & Governance*, 15(3), 561–580. https:// doi.org/10.1111/rego.12243

- Kutlina-Dimitrova, Z., & Lakatos, C. (2016). Determinants of direct cross-border public procurement in EU member states. *Review of World Economics*, 152(3), 501–528. https://doi.org/10.1007/s10290-016-0251-3
- Lakka, S., Stamati, T., Michalakelis, C., & Martakos, D. (2013). What drives eGovernment growth? An econometric analysis on the impacting factors. *International Journal of Electronic Governance*, 6(1), 20–36. https://doi.org/10.1504/1JEG.2013.053366

Larbi, G. A. (2007). Between spin and reality: Examining disclosure practices in three African countries. Public Administration and Development, 27(3), 205–214. https:// doi.org/10.1002/pad.451

Lathrop, D., & Ruma, L. (2010). Open government: Collaboration, transparency, and participation in practice. "O'Reilly Media, Inc."

Lenk, K. (2006). E-government in Europe: Uniform solutions for all countries? Information Polity, 11(3-4), 189–196.

Levy, B., & Spiller, P. T. (1994). The institutional foundations of regulatory commitment: A comparative analysis of telecommunications regulation. *The Journal of Law, Economics, and Organization, 10*(2), 201–246. https://doi.org/10.1093/ oxfordjournals.ileo.a036849

Lewis-Faupel, S., Neggers, Y., Olken, B. A., & Pande, R. (2016). Can electronic procurement improve infrastructure provision? Evidence from public works in India and Indonesia. *American Economic Journal: Economic Policy*, 8(3), 258–283.

Lidén, G. (2015). Technology and democracy: Validity in measurements of e-democracy. Democratization, 22(4), 698–713. https://doi.org/10.1080/13510347.2013.873407

Linarelli, J. (2003). The WTO transparency agenda: Law, economics and international relations theory. In S. Arrowsmith, & M. Trybus (Eds.), *Public procurement: The continuing revolution* (pp. 235–268). MA: Kluwer Law International.

Lnenicka, M., Luterek, M., & Nikiforova, A. (2022). Benchmarking open data efforts through indices and rankings: Assessing development and contexts of use. *Telematics* and Informatics, 66, Article 101745. https://doi.org/10.1016/j.tele.2021.101745

Lnenicka, M., Nikiforova, A., Luterek, M., Milic, P., Rudmark, D., Neumaier, S., ... Bolívar, M. P. R. (2024). Identifying patterns and recommendations of and for sustainable open data initiatives: A benchmarking-driven analysis of open government data initiatives among European countries. *Government Information Quarterly*, 41(1), Article 101898. https://doi.org/10.1016/j.gig.2023.101898

- Lourenço, R. P. (2015). An analysis of open government portals: A perspective of transparency for accountability. *Government Information Quarterly*, 32(3), 323–332. https://doi.org/10.1016/j.giq.2015.05.006
- Lourenço, R. P. (2023). Government transparency: Monitoring public policy accumulation and administrative overload. Government Information Quarterly, 40(1), Article 101762. https://doi.org/10.1016/j.giq.2022.101762

Madan, R., & Ashok, M. (2023). AI adoption and diffusion in public administration: A systematic literature review and future research agenda. *Government Information Quarterly*, 40(1), Article 101774. https://doi.org/10.1016/j.giq.2022.101774 Margetts, H. (2011). The internet and transparency. *The Political Quarterly*, 82(4), 518–521.

- Matheus, R., Janssen, M., & Maheshwari, D. (2020). Data science empowering the public: Data-driven dashboards for transparent and accountable decision-making in smart cities. Government Information Quarterly, 37(3), 101284. https://doi.org/10.1016/j. gia.2018.01.006
- Meijer, A. (2013). Understanding the complex dynamics of transparency. Public Administration Review, 73(3), 429–439. https://doi.org/10.1111/puar.12032
- Meijer, A. (2015). Government transparency in historical perspective: From the ancient regime to open data in the Netherlands. *International Journal of Public Administration*, 38(3), 189–199. https://doi.org/10.1080/01900692.2014.934837
- Meijer, A. J., Curtin, D., & Hillebrandt, M. (2012). Open government: Connecting vision and voice. International Review of Administrative Sciences, 78(1), 10–29. https://doi. org/10.1177/0020852311429533

Mélon, L., & Spruk, R. (2020). The impact of e-procurement on institutional quality. Journal of Public Procurement, 20(4), 333–375. https://doi.org/10.1108/JOPP-07-2019-0050

Min, H., & Galle, W. P. (2003). E-purchasing: Profiles of adopters and nonadopters. Industrial Marketing Management, 32(3), 227–233. https://doi.org/10.1016/S0019-8501(02)00266-3

Molander, P. (2014). Public procurement in the European Union: The case for National Threshold Values. Journal of Public Procurement, 14(2), 181–214. https://doi.org/ 10.1108/JOPP-14-02-2014-B002

Moon, M. J. (2005). E-procurement management in state governments: Diffusion of eprocurement practices and its determinants. *Journal of Public Procurement*. https:// doi.org/10.1108/JOPP-05-01-2005-B003

Moser-Plautz, B. (2023). Barriers to digital government and the COVID-19 crisis–a comparative study of federal government entities in the United States and Austria. *International Review of Administrative Sciences*. https://doi.org/10.1177/ 00208523231183566

Mossberger, K., Tolbert, C. J., & Stansbury, M. (2003). Virtual inequality: Beyond the digital divide. Washington, DC: Georgetown University Press.

Mungiu-Pippidi, A. (2023). Transparency and corruption: Measuring real transparency by a new index. *Regulation & Governance*, 17, 1094–1113. https://doi.org/10.1111/ rego.12502

Mungiu-Pippidi, A., & Dadašov, R. (2016). Measuring control of corruption by a new index of public integrity. European Journal on Criminal Policy and Research, 22, 415–438. https://doi.org/10.1007/s10610-016-9324-z

Nizalova, O. Y., & Murtazashvili, I. (2016). Exogenous treatment and endogenous factors: Vanishing of omitted variable bias on the interaction term. *Journal of Econometric Methods*, 5(1), 71–77. https://doi.org/10.1515/jem-2013-0012

North, D. C. (1990). Institutions, institutional change and economic performance. New York N.Y: Cambridge University Press.

S. Khorana et al.

Noveck, B. (2009). Wiki government: How technology can make government better, democracy stronger, and citizens more powerful. Washington, DC: Brookings Institution Press.

OECD. (2008). Handbook on constructing composite indicators: Methodology and user guide. OECD.

OECD. (2016). Open government: The global context and the way forward. Paris: OECD Publishing. https://doi.org/10.1787/9789264268104-en

OECD. (2017). Government at a glance 2017. Paris: OECD Publishing. https://doi.org/ 10.1787/gov_glance-2017-en

OECD. (2018). Methodology for assessing procurement systems (MAPS). Paris: OECD Publishing. Online access: https://www.mapsinitiative.org/methodology/MAPS-Methodology-ENG.pdf.

OECD. (2019a). Government at a glance 2019. Paris: OECD Publishing. https://doi.org/ 10.1787/8ccf5c38-en

OECD. (2019b). Reforming public procurement: Progress in implementing the 2015 OECD recommendation. OECD Public Governance Reviews. Paris: OECD Publishing.

OECD. (2019c). Promoting transparency and strategic use of public procurement. Online access: https://www.oecd.org/corruption-integrity/checklists/promoting-effic ient-transparent-sub-national-procurementaci.html.

OECD. (2020). Fair market conditions for competitiveness in the adriatic region policy briefing. Available at: https://www.oecd.org/south-east-europe/programme/ OECD-Public-Procurement-Policy-Briefing-Note.pdf.

OECD. (2020a). Open, useful and re-usable data (OURdata) index: 2019. In OECD public governance policy papers, No. 01. Paris: OECD Publishing https://doi.org/10.1787/ 45f6de2d-en.

OECD. (2021). Government at a glance 2021. Paris: OECD Publishing. https://doi.org/ 10.1787/1c258f55-en

O'Shaughnessy, M. R., Schiff, D. S., Varshney, L. R., Rozell, C. J., & Davenport, M. A. (2023). What governs attitudes toward artificial intelligence adoption and governance? *Science and Public Policy*, 50(2), 161–176. https://doi.org/10.1093/ scipol/scac056

Otenyo, E. E., & Lind, N. S. (2004). Faces and phases of transparency reform in local government. International Journal of Public Administration, 27(5), 287–307. https:// doi.org/10.1081/PAD-120028811

Pani, A. K., & Kar, A. K. (2011). A study to compare relative importance of criteria for supplier evaluation in e-procurement. In 2011 44th hawaii international conference on system sciences (pp. 1–8). IEEE.

- Papke, L. E., & Wooldridge, J. M. (1996). Econometric methods for fractional response variables with an application to 401 (k) plan participation rates. *Journal of Applied Econometrics*, 11(6), 619–632. https://doi.org/10.1002/(SICI)1099-1255(199611) 11:6{619::AID-JAE418 near here}3.0.CO;2-1
- Pathak, R. D., Naz, R., Rahman, M. H., Smith, R. F. I., & Nayan Agarwal, K. (2009). Egovernance to cut corruption in public service delivery: A case study of fiji. *Intl Journal of Public Administration*, 32(5), 415–437. https://doi.org/10.1080/ 01900690902799482
- Pathak, R. D., Singh, G., Belwal, R., Naz, R., & Smith, R. F. I. (2008). E-governance, corruption and public service delivery: A comparative study of Fiji and Ethiopia. *Journal of Administration and Governance*, 3(1), 65–79.
- Peterson, R. A. (1994). A meta-analysis of Cronbach's coefficient alpha. Journal of Consumer Research, 21(2), 381–391. https://doi.org/10.1086/209405
- Pirannejad, A., Janssen, M., & Rezaei, J. (2019). Towards a balanced E-participation index: Integrating government and society perspectives. *Government Information Quarterly*, 36(4), Article 101404. https://doi.org/10.1016/j.giq.2019.101404
- Pyman, M., & Heywood, P. (2021). The sector focus & reformulation approach (SFRA). In *Curbing corruption and global integrity*. Washington DC Global Integrity-ACE Working Paper 5.
- Rai, A., Tang, X., Brown, P., & Keil, M. (2006). Assimilation patterns in the use of electronic procurement innovations: A cluster analysis. *Information & Management*, 43(3), 336–349. https://doi.org/10.1016/j.im.2005.08.005
- Relly, J. E., & Sabharwal, M. (2009). Perceptions of transparency of government policymaking: A cross-national study. *Government Information Quarterly*, 26(1), 148–157. https://doi.org/10.1016/j.giq.2008.04.002

Rezaei, J. (2015). Best-worst multi-criteria decision-making method. Omega, 53, 49–57. https://doi.org/10.1016/j.omega.2014.11.009

Rezaei, J. (2016). Best-worst multi-criteria decision-making method: Some properties and a linear model. Omega, 64, 126–130. https://doi.org/10.1016/j. omega.2015.12.001

Roberts, A. (2006). Blacked out: Government secrecy in the information age. Cambridge University Press.

Rothstein, B. (2014). What is the opposite to corruption? *Third World Quarterly*, 35(5), 737–752. https://doi.org/10.1080/01436597.2014.921424

Ruiz Rivadeneira, A., Dekyi, T., & Cruz, L. (2023). OECD infrastructure governance indicators: Conceptual framework, design, methodology and preliminary results. In OECD working papers on public governance, No. 59. Paris: OECD Publishing. https:// doi.org/10.1787/95c2cef2-en.

Savoldelli, A., Codagnone, C., & Misuraca, G. (2014). Understanding the e-government paradox: Learning from literature and practice on barriers to adoption. *Government Information Quarterly*, 31, S63–S71. https://doi.org/10.1016/j.giq.2014.01.008

Schapper, P. (2008). The impact of e-procurement on corruption: The potential of eprocurement for curbing corruption risks. In ADB/OECD (Ed.), *Fighting bribery in public procurement in Asia and the Pacific* (pp. 113–128). Paris: OECD Publishing.

Schmidthuber, L., Ingrams, A., & Hilgers, D. (2021). Government openness and public trust: The mediating role of democratic capacity. *Public Administration Review*, 81(1), 91–109. https://doi.org/10.1111/puar.13298 Schopf, J. C. (2019). Room for improvement: Why Korea's leading ICT ODA program has failed to combat corruption. *Telecommunications Policy*, 43(6), 501–519. https://doi. org/10.1016/j.telpol.2019.01.001

Schumacker, R. E., & Lomax, R. G. (2004). A beginner's guide to structural equation modeling. Psychology Press.

Schwab, B., Janzen, S., Magnan, N. P., & Thompson, W. M. (2020). Constructing a summary index using the standardized inverse-covariance weighted average of indicators. *The Stata Journal*, 20(4), 952–964. https://doi.org/10.1177/ 1536867X20976325

Searson, E. M., & Johnson, M. A. (2010). Transparency laws and interactive public relations: An analysis of latin american government web sites. *Public Relations Review*, 36(2), 120–126. https://doi.org/10.1016/j.pubrev.2010.03.003

Seong, S. K., & Lee, J. Y. (2004). Developing e-procurement systems: A case study on the government e-procurement systems in Korea. *Public Finance & Management*, 4(2), 138–166.

Shapiro, A. L. (1999). The control revolution: How the internet is putting individuals in charge and changes the world. New York: Public Affairs.

Shim, D. C., & Eom, T. H. (2008). E-government and anti-corruption: Empirical analysis of international data. International Journal of Public Administration, 31(3), 298–316. https://doi.org/10.1080/01900690701590553

Shingal, A. (2011). Services procurement under the WTO's agreement on government procurement: Whither market access? World Trade Review, 10(4), 527–549. https:// doi.org/10.1017/S1474745611000309

Srivastava, S. C., & Teo, T. S. (2010). E-government, e-business, and national economic performance. Communications of the Association for Information Systems, 26(1), 14. https://doi.org/10.17705/1CAIS.02614

Standing, C., Love, P. E., Stockdale, R., & Gengatharen, D. (2006). Examining the relationship between electronic marketplace strategy and structure. *IEEE Transactions on Engineering Management*, 53(2), 297–311. https://doi.org/10.1109/ TEM.2005.861801

Stasavage, D. (2003). Transparency, democratic accountability, and the economic consequences of monetary institutions. *American Journal of Political Science*, 47(3), 389–402. https://doi.org/10.1111/1540-5907.00028

Stevens, J. (2002). Applied multivariate statistics for the social sciences, 4. Mahwah, NJ: Lawrence Erlbaum Associates.

Szepesi, S. (2004). Coercion or engagement? Economics and institutions in ACP-EU trade negotiations. In , 56. Economics and institutions in ACP/EU trade negotiations (June 2004). European Centre for Development Policy Management Discussion Paper.

Tornatzky, L. G., & Fleischer, M. (1990). *The processes of technological innovation*. Lexington, MA: Lexington Book.

Trepte, P. (2005). Transparency and accountability as tools for promoting integrity and preventing corruption in procurement: Possibilities and limitations. OECD Papers, 5 (10), 1–37.

Twizeyimana, J. D., & Andersson, A. (2019). The public value of E-government–A literature review. Government Information Quarterly, 36(2), 167–178. https://doi. org/10.1016/j.gig.2019.01.001

United Nations. (2011). Transparency and public procurement: Supplement to the 2011 annual statistical report on united nations procurement. Available: https://www. undp.org/sites/g/files/zskgke326/files/undp/library/corporate/Executive%20Bo ard/2012/second-regular-session/English/2011_supplement_transparency.pdf.

Varney, M. (2011). E-procurement—Current law and future challenges, in ERA forum. 12 pp. 185–204). Springer. https://doi.org/10.1007/s12027-011-0217-9

Vinogradov, D., Shadrina, E., & Kokareva, L. (2014). Public procurement mechanisms for public-private partnerships. Journal of Public Procurement. https://doi.org/10.1108/ JOPP-14-04-2014-B004

Wang, H. J., & Lo, J. (2016). Adoption of open government data among government agencies. Government Information Quarterly, 33(1), 80–88. https://doi.org/10.1016/ j.giq.2015.11.004

Ware, G. T., Moss, S., Campos, J. E., & Noone, G. P. (2007). Corruption in public procurement: A perennial challenge. In J. E. Campos, & S. Pradhan (Eds.), *The many faces of corruption. Tracking vulnerabilities at the sector level.* Washington, DC: World Bank.

Whitley, R. (1999). Divergent capitalisms: The social structuring and change of business systems. Oxford: Oxford University Press.

Williams, A. (2009). On the release of information by governments: Causes and consequences. Journal of Development Economics, 89(1), 124–138. https://doi.org/ 10.1016/j.jdeveco.2008.08.001

Williams, A. (2011). Shining a light on the resource curse: An empirical analysis of the relationship between natural resources, transparency, and economic growth. World Development, 39(4), 490–505. https://doi.org/10.1016/j.worlddev.2010.08.015

Williams, A. (2015). A global index of information transparency and accountability. Journal of Comparative Economics, 43(3), 804–824. https://doi.org/10.1016/j. jce.2014.10.004

Williams, M. D. J., Mayer, R., & Minges, M. (2011). Africa's ICT infrastructure: Building on the mobile revolution. Washington, DC: World Bank.

Wolfe, R. (2013). Letting the sunshine in at the WTO: How transparency brings the trading system to life. In WTO staff working paper ERSD-2013-03.

Wooldridge, J. M. (2010). Econometric analysis of cross section and panel data (2nd ed.). The MIT Press.

World Bank. (2013). Global stock-take of social accountability initiatives for budget transparency and monitoring: Key challenges and lessons learned. In World bank publications - reports 16661. The World Bank Group. Online access: https://openk nowledge.worldbank.org/bitstream/handle/10986/16661/815430REVISED00ockt ake0ReportOFinal.pdf?sequence=5.

World Bank. (2016). World development report 2016: Digital dividends. Washington D.C.

S. Khorana et al.

- World Bank. (2023). World development indicators. Available at: https://data.worl dbank.org [accessed 27 May 2023].
- World Trade Organization (WTO). (2012). The plurilateral agreement on government procurement (GPA) (Geneva: WTO). Online access: https://www.wto.org/english/ docs_e/legal_e/rev-gpr-94_01_e.pdf.
- World Trade Organization (WTO). (2023). WTO's contribution to attaining UN sustainable development goals: Update to the high-level political forum. Online access: https://www.wto.org/english/res_e/booksp_e/un_hlpf23_e.pdf.
- Wyld, D. C. (2002). The electric company: How the supply chain is being reinvented through the rapid application of e-procurement processes in the business-to-business arena. *Management Research News*, 25(12), 22–53. https://doi.org/10.1108/ 01409170210783269
- Zheng, Y. (2016). The impact of E-participation on corruption: A cross-country analysis. International Review of Public Administration, 21(2), 91–103. https://doi.org/ 10.1080/12294659.2016.1186457
- Zhu, Y., Li, Y., Wang, W., & Chen, J. (2010). What leads to post-implementation success of ERP? An empirical study of the chinese retail industry. *International Journal of Information Management*, 30(3), 265–276. https://doi.org/10.1016/j. ijinfomgt.2009.09.007

Sangeeta Khorana is a Professor of International Trade Policy at the Aston Business School at Aston University, UK. She is an international trade expert with 25+ years of academia and management consulting experience including, between 1993 and 2001 as a civil servant with the government of India. Prof Khorana leads research on international trade policy, in particular on free trade agreement (FTA) negotiations and WTO issues. She has advised EU institutions on FTA negotiations with India and Vietnam as well as developed methodologies for liberalisation through FTA negotiations. Prof Khorana has undertaken funded research for the UK Economic and Social Research Council (ESRC), Welsh Assembly, Foreign & Commonwealth Office, European Commission, DG Trade, Commonwealth Secretariat, InterAmerican Development Bank (IADB), World Bank-ITCILO, UNCTAD-India and the Swiss Federal Government among others.

Santiago Caram is an Applied Economist. He earned his PhD in Economics, Management and Statistics from UNIME in Italy. Before that he received an MA in Economics from UCEMA and a BS in Economics from UNLP in Argentina. He has more than nine years of experience in both private and public sectors. His duties were related to econometrics models, trackers and tax analysis. His research interests include Development Economics, Applied Economics and Labour Economics. He is particularly interested on the developmental effects of infrastructure expenditures, public procurement, preferential trade agreements and environmental issues over labour markets.

Nripendra P. Rana* is a Professor in Digital Marketing and Systems at the Queen's Business School, Queen's University Belfast, UK. His current research interests focus primarily on adoption and diffusion of emerging ICTs, e-commerce, m-commerce, digital and social media marketing and the role of artificial intelligence on the consumer decisionmaking and behaviour. He has published more than 350 papers in a range of leading academic journals, conference proceedings, books etc. Some of such journals where his papers got published include British Journal of Management, European Journal of Information Systems, Annals of Tourism Research, Public Management Review, European Journal of Information Management to name a few. He is an Associate Editor of International Journal of Infor-